PACKINGS OF PARTIAL DIFFERENCE SETS

Jonathan Jedwab^{*1} and Shuxing Li^{†2}

^{1,2}Department of Mathematics, Simon Fraser University, 8888 University Drive, Burnaby BC V5A 1S6, Canada jed@sfu.ca, shuxing_li@sfu.ca

> Submitted: Apr 28, 2021; Accepted: Sep 9, 2021; Published: TBD © The authors. Released under the CC BY license (International 4.0).

ABSTRACT. A packing of partial difference sets is a collection of disjoint partial difference sets in a finite group G. This configuration has received considerable attention in design theory, finite geometry, coding theory, and graph theory over many years, although often only implicitly. We consider packings of certain Latin square type partial difference sets in abelian groups having identical parameters, the size of the collection being either the maximum possible or one smaller. We unify and extend numerous previous results in a common framework, recognizing that a particular subgroup reveals important structural information about the packing. Identifying this subgroup allows us to formulate a recursive lifting construction of packings in abelian groups of increasing exponent, as well as a product construction yielding packings in the direct product of the starting groups. We also study packings of certain negative Latin square type partial difference sets of maximum possible size in abelian groups, all but one of which have identical parameters, and show how to produce such collections using packings of Latin square type partial difference sets. **Keywords:** Finite abelian group, packing, partial difference set

Mathematics Subject Classifications: 05B10, 20K01

1. Introduction

A recurring theme across many diverse areas of mathematics is the natural occurrence of structures for which a key aspect can take one of only two values. We mention several examples of this two-valued phenomenon. In coding theory, a projective two-weight code is a linear code whose codeword weights take one of exactly two distinct values [13]. In finite geometry, a projective two-intersection set is a point set in projective space whose intersection with each hyperplane has one of exactly two distinct sizes [13]; and an m-ovoid or a tight set in a polar space is a point set whose intersection with every tangent hyperplane of the polar space has one of exactly two distinct sizes [10, Chapter 2], [66, Section 4.5]. In graph theory, a strongly regular graph has

^{*}supported by an NSERC Discovery Grant.

[†]supported by a PIMS Postdoctoral Fellowship.

exactly two distinct eigenvalues [10, Chapter 1]. In group theory, a rank 3 permutation group has exactly two nontrivial orbitals [13, Section 10]; and a symmetric Schur ring of dimension 3 over a group G is a partition of the nonidentity group elements into exactly two nontrivial subsets which, together with the identity element, form a subalgebra of $\mathbb{C}[G]$ [90]. In number theory, uniform cyclotomy is a partition of the nonzero elements of a finite field into cyclotomy classes such that exactly two distinct cyclotomic numbers occur [6].

An underlying connection between many instances of these two-valued phenomena is provided by a partial difference set. Indeed, let D be a k-subset of an additive abelian group G of order v. The subset D is a (v, k, λ, μ) partial difference set in G if the multiset $\{x - y \mid x, y \in D, x \neq y\}$ contains each nonidentity element of D exactly λ times and each nonidentity element of $G \setminus D$ exactly μ times. Partial difference sets, possibly having some additional properties, give rise to examples of each of the above structures. The construction of partial difference sets is therefore of great interest. We refer to [13, 60] for excellent surveys of partial difference sets and equivalent structures, and to [1, 2, 4, 5, 8, 9, 11, 12, 14, 15, 17, 18, 19, 20, 21, 22, 24, 25, 26, 27, 28, 29, 30, 31, 32, 37, 38, 42, 43, 44, 45, 46, 47, 48, 49, 50, 51, 52, 54, 58, 59, 61, 62, 63, 64, 65, 67, 68, 69, 70, 71, 72, 73, 74, 75, 76, 77, 78, 79, 80, 82, 84, 85, 86, 89] for investigations spanning twenty-five years into which groups G contain a partial difference set with specified parameters (v, k, λ, μ) .

This paper is concerned with the construction of two of the richest classes of partial difference sets, whose parameters are determined by only two integers. A partial difference set whose parameters (v, k, λ, μ) take the form $(n^2, r(n-1), n + r(r-3), r(r-1))$ has (n, r) Latin square type, and one whose parameters take the form $(n^2, r(n+1), -n + r(r+3), r(r+1))$ has (n,r) negative Latin square type. A partial difference set D in G is regular if $1_G \notin D$ and D = $\{d^{-1} \mid d \in D\}$. A great deal of research has been conducted into the existence of a collection of t > 1 disjoint regular (tc, c) Latin square type partial difference sets in an abelian group G of order t^2c^2 . The number of elements of G avoided by the partial difference sets of such a collection is tc, so the collection has maximum possible size when c > 1. The principal motivation for this paper was the discovery of a widespread structural property that does not seem to have been previously recognized: in all previous constructions, possibly after some modification, the tc avoided elements of G form a subgroup U. We therefore refer to such a collection of partial difference sets as a (c, t) LP-packing in G relative to U. The presence of the associated subgroup U is key to the recursive lifting construction of LP-packings of increasing exponent that we present in Section 5. In general, lifting constructions (that increase the group exponent) have historically been more challenging to find than product constructions (that combine objects into the direct product of the starting groups). Table 1.1 illustrates how widely the concept of LPpackings has been previously studied, often implicitly, and in each case identifies the subgroup Uexplicitly. In some cases, identifying the crucial subgroup U from the original reference requires considerable effort.

We shall unify and extend many results for LP-packings, whose original derivation relied on a variety of sometimes delicate approaches, by means of a common framework that depends only on elementary methods. We shall show that we have considerable control over the choice of the associated subgroup U. In addition to the recursive construction of Section 5, we shall give a product construction for LP-packings in Section 4. We present the following result (to be Combinatorial theory 27 (2020), #P00

proved as Corollary 5.4) as a showcase for our constructions.

Theorem 1.1. Let p be prime, let s_1, \ldots, s_v be nonnegative integers (not all zero), and let $m = \min\{s_i \mid s_i > 0\}$. For each $i = 1, \ldots, v$, let $G_i = \mathbb{Z}_{p^i}^{2s_i}$ and let U_i be a subgroup of G_i of order p^{is_i} . Let $G = \prod_{i=1}^v G_i$, let $U = \prod_{i=1}^v U_i$, and let $n = \sqrt{|G|}$. Then for each $j = 0, \ldots, m-1$, there exists an $\left(\frac{n}{p^{m-j}}, p^{m-j}\right)$ LP-packing in G relative to U: a collection of p^{m-j} disjoint regular $\left(n, \frac{n}{p^{m-j}}\right)$ Latin square type partial difference sets in G avoiding U.

We shall show in Section 3 that a collection of t disjoint regular (tc, c) Latin square type partial difference sets with c = 1 does not have maximum possible size, because there exists a size t+1 collection known equivalently as a (t, t+1) partial congruence partition. However, we regard the (1, t) LP-packing as a more natural object than the larger partial congruence partition, because it forms the base case of the recursive construction of Section 5 and so allows a powerful generalization to nonelementary abelian groups.

We similarly refer to a collection of t - 1 > 0 disjoint regular (tc, c) negative Latin square type partial difference sets in an abelian group G of order t^2c^2 , for which the (c - 1)(tc + 1)nonidentity avoided elements of G form a regular (tc, c - 1) negative Latin square type partial difference set in G, as a (c, t - 1) *NLP-packing*; such a collection has maximum possible size for all $c \ge 1$. The NLP-packing structure was already known, and previous constructions are summarized in Table 1.2. We shall use a product construction in Section 6 to combine small examples of NLP-packings with the families of LP-packings. We mention that both LP-packings and NLP-packings give rise to group-based amorphic association schemes [87].

Our principal objective is to determine for which abelian groups G and subgroups U there exists a (c, t) LP-packing in G relative to U, and for which abelian groups G there exists a (c, t-1) NLP-packing in G. We propose that the natural framework for studying regular (tc, c) Latin square type and negative Latin square type partial difference sets is as these respective collections, rather than as single examples.

The rest of the paper is organized as follows. Section 2 gives a historical overview of packings of partial difference sets of Latin square type and negative Latin square type in abelian groups. Section 3 introduces LP-packings, describes their relationship to partial congruence partitions, establishes constraints on their structure, and presents a product construction. Section 4 introduces an LP-partition as an auxiliary configuration in the construction of an LP-packing in a larger group from an LP-packing in a smaller group. Section 5 recursively constructs infinite families of LP-partitions and LP-packings of increasing exponent, inspired by a recursive lifting construction of difference sets using relative difference sets [33]. Section 6 introduces NLP-packings, and uses a product construction to combine an NLP-packing with an LP-packing to give an NLP-packing in the direct product of the starting groups. Section 7 examines how LP-packings are related to hyperbolic and elliptic strongly regular bent functions, and Section 8 examines how these packings are related to reversible Hadamard difference sets. Section 9 proposes some open problems for future research.

	С	t	Group G	Subgroup U	Restrictions	Source
	p^{w-s}	p^s	$\mathbb{Z}_{p^2}^w$	$(p\mathbb{Z}_{p^2})^w$	s is proper divisor of $w > 1$	[21, Thm. 3.1]
	p	p	$\mathbb{Z}_{p^2}^2$	$(p\mathbb{Z}_{p^2})^2$		[32 , Thm. 3.1]
	2^{a-1}	2	$\mathbb{Z}_{2^a}^2$	order 2^a , noncyclic		[36 , Thm. 3.1]
	$p^{w-\sigma(p,w)}$	$p^{\sigma(p,w)}$	$\mathbb{Z}_{p^2}^w$	$(p\mathbb{Z}_{p^2})^w$	w > 1	[51, Lem. 3.1]
	$\frac{n}{p^s}$	p^s	$\mathbb{Z}_{p^{a+1}}^{2us} imes \mathbb{Z}_{p^a}^{2ws}$	$\mathbb{Z}_{p^{a+1}}^{us} imes \mathbb{Z}_{p^a}^{ws}$	$n = \sqrt{ G }$ and $a \ge 1$	[57, Thm. 3.1]
	p^{w-1}	p	$\mathbb{Z}_{p^2}^w$	$(p\mathbb{Z}_{p^2})^w$	w > 1	[59, Thm. 2.2]
	p^s	p^s	$\mathbb{Z}_{p^2}^{2s}$	$(p\mathbb{Z}_{p^2})^{2s}$	s > 1	[80, Prop. 4.1]
-	$p^{(2a-1)s}$	p^s	$\mathbb{Z}_{p^{2a}}^{2s}$	$(p^a \mathbb{Z}_{p^{2a}})^{2s}$		[81, Lem. 4.3]
						[82, Thm. 3.1]
	$\frac{n}{p}$	p	$\mathbb{Z}_p^{2u} \times \prod_{i=1}^v \mathbb{Z}_{p^{2i}}^{2s_{2i}}$	$\mathbb{Z}_p^u \times \prod_{i=1}^v (p^i \mathbb{Z}_{p^{2i}})^{2s_{2i}}$	$n = \sqrt{ G } \text{ and }$ $u, s_{2i} \ge 0$	[82, Cor. 6.2]
	p^{w-1}	p	$\mathbb{Z}_{p^2}^w$	$(p\mathbb{Z}_{p^2})^w$	p is odd divisor of w	[84, Thm. 4.1]
	$2^{\lceil w/2 \rceil}$	$2^{\lfloor w/2 \rfloor}$	\mathbb{Z}_4^w	$(2\mathbb{Z}_4)^w$	w > 1	[91, Lem. 4.5]

Table 1.1: Known (c, t) LP-packings in G relative to U, where p is prime and σ is defined in (2.2)

2. Historical Overview

In this section, we give an overview of previous constructions of regular Latin square type and negative Latin square type partial difference sets in abelian groups. In many cases, these results produce only a single partial difference set, whereas our objective in later sections will be to construct disjoint collections.

Let G be an abelian group. We firstly present some basic results involving the group ring $\mathbb{Z}[G]$ and character theory. For $A \in \mathbb{Z}[G]$ and $g \in G$, denote the coefficient of g in A by $[A]_g$ (so that $A = \sum_{g \in G} [A]_g g$). For $A \in \mathbb{Z}[G]$, write $A^{(-1)} = \sum_{g \in G} [A]_g g^{-1}$. For a subset A of G, by a standard abuse of notation we also denote the group ring element $\sum_{g \in A} g$ by A. A *character* χ of G is a group homomorphism from G to the multiplicative group of the complex field \mathbb{C} . Write \hat{G} for the character group of G. A character $\chi \in \hat{G}$ is principal on a subgroup $H \leq G$ if $\chi(h) = 1$ for each $h \in H$. For a subgroup $H \leq G$, write

$$H^{\perp} = \{ \chi \in \widehat{G} \mid \chi \text{ is principal on } H \}$$

(so that G^{\perp} consists of only the principal character on G). For $\chi \in \widehat{G}$ and $A \in \mathbb{Z}[G]$, write $\chi(A)$ for the character sum $\sum_{g \in G} [A]_g \chi(g)$. For a more detailed treatment of group rings and character theory, see [7, Chap. VI] and [83, Chap. 1].

Definition 2.1. Let *D* be a *k*-subset of an additive group *G* of order *v*. The subset *D* is a (v, k, λ, μ) partial difference set (PDS) in *G* if the multiset $\{x - y \mid x, y \in D, x \neq y\}$ contains

с	t - 1	Group G	Restrictions	Source
$\frac{n}{2}$	1	$\mathbb{Z}_2^{2u} \times \mathbb{Z}_4^w \times \prod_{i=3}^v \mathbb{Z}_{2^i}^{2s_i} \times \mathbb{Z}_3^{2y} \times \prod_{i=1}^z \mathbb{Z}_{p_i}^{4t_i}$	$n = \sqrt{ G }$ and prime $p_i > 3$, $u, w, s_i, y, t_i \ge 0$,	[20, Prop. 3.12]
			if $u = 0$ then $y = t_i = 0$	
$\frac{n}{3}$	2	$\mathbb{Z}_3^{2+2u} \times \mathbb{Z}_9^w \times \prod_{i=3}^v \mathbb{Z}_{3^i}^{2s_i}$	$n = \sqrt{ G } \text{ and } u, w, s_i \ge 0,$ $w \ne 1$	[20, Prop. 3.13]
<u>n</u>	3	$\mathbb{Z}_2^{4+2u} imes \mathbb{Z}_4^w imes \prod_{i=3}^v \mathbb{Z}_{2^i}^{4s_i}$	$n = \sqrt{ G }$ and $u, w, s_i \ge 0$,	[20 Prop 3.14]
4	5	$\mathbb{Z}_2^{2u} \times \mathbb{Z}_4^{w+2} \times \prod_{i=3}^v \mathbb{Z}_{2^i}^{4s_i}$	$u \neq 1 \text{ and } w \notin \{1,3\}$	[20, 110p. 5.14]
<i>⊿w−</i> 1	2	$\mathbb{Z}_4^{2u} \times \mathbb{Z}_2^{4w-4u}$	$0 \leqslant u \leqslant w \text{ if } w \text{ odd}$	[38, Cor. 2.2]
4	0		$0 \leqslant u < w$ if w even	[39 , Thm. 3.6]
$\frac{n}{4}$	3	$\mathbb{Z}_2^{4u} \times \mathbb{Z}_4^{2w} \times \prod_{i=2}^v \mathbb{Z}_{2^{2i}}^{4s_{2i}}$	$n = \sqrt{ G } \text{ and } u, w, s_{2i} \ge 0,$ $u + w \ge 1$	[82, Thm. 4.2]
$\frac{n}{3}$	2	$\mathbb{Z}_{3}^{2u+2} \times \prod_{i=1}^{v} \mathbb{Z}_{3^{2i}}^{2s_{2i}}$	$n = \sqrt{ G }$ and $u, s_{2i} \ge 0$	[82, Cor. 5.1]
3^{w-1}	2	\mathbb{Z}_3^{2w}	$w \ge 1$	[86, Thm. 1]

Table 1.2: Known (c, t - 1) NLP-packings in G

each nonidentity element of D exactly λ times and each nonidentity element of G - D exactly μ times. The partial difference set is *regular* if $1_G \notin D$ and $D = D^{(-1)}$.

The condition $D = D^{(-1)}$ in Definition 2.1 is guaranteed to hold except in the special case $\lambda = \mu$ [60, Prop. 1.2], in which case the partial difference set is a (v, k, λ) -difference set in G (see Section 8). Provided that $D = D^{(-1)}$, the condition $1_G \notin D$ is not restrictive [60, p. 222]; we shall be concerned only with partial difference sets that are regular. Let D be a k-subset of a group G of order v for which $1_G \notin D$. Then, in group ring notation, D is a (v, k, λ, μ) PDS in G if and only if

$$DD^{(-1)} = k - \mu + (\lambda - \mu)D + \mu G \quad \text{in } \mathbb{Z}[G].$$
(2.1)

The following result is a consequence of the orthogonality properties of characters.

Proposition 2.2 (Fourier inversion formula). Let G be an abelian group and let $A \in \mathbb{Z}[G]$. Then

$$[A]_g = \frac{1}{|G|} \sum_{\chi \in \widehat{G}} \chi(A) \overline{\chi(g)} \quad \text{for each } g \in G.$$

In particular, elements A, B of $\mathbb{Z}[G]$ are equal if and only if $\chi(A) = \chi(B)$ for all $\chi \in \widehat{G}$.

For a regular PDS D, we can use the relation $D = D^{(-1)}$ to rewrite (2.1) as $D^2 = k - \mu + (\lambda - \mu)D + \mu G$ in $\mathbb{Z}[G]$. Applying a nonprincipal character χ to both sides then shows that the character sum $\chi(D)$ satisfies the quadratic equation $\chi(D)^2 - (\lambda - \mu)\chi(D) - (k - \mu) = 0$. Using Proposition 2.2, we therefore obtain the following characterization of a regular partial difference set in terms of its character sums.

Lemma 2.3 ([60, Cor. 3.3]). Let G be an abelian group of order v. Let D be a k-subset of G for which $1_G \notin D$. Let λ, μ be nonnegative integers satisfying $k^2 = k - \mu + (\lambda - \mu)k + \mu v$ and $(\lambda - \mu)^2 + 4(k - \mu) \ge 0$. Then D is a regular (v, k, λ, μ) partial difference set in G if and only if

$$\chi(D) = \frac{1}{2} \left(\lambda - \mu \pm \sqrt{(\lambda - \mu)^2 + 4(k - \mu)} \right) \quad \text{for all nonprincipal characters } \chi \text{ of } G.$$

We are interested in the following two types of parameters for partial difference sets.

Definition 2.4. Let $n \ge 1$ and $r \ge 0$ be integers. An $(n^2, r(n - \epsilon), \epsilon n + r^2 - 3\epsilon r, r^2 - \epsilon r)$ PDS is an (n, r) Latin square type PDS if $\epsilon = 1$, and is an (n, r) negative Latin square type PDS if $\epsilon = -1$.

The connection between Latin square type partial difference sets and Latin squares is "rather indirect" [52, p. 2]. We allow the case r = 0 in Definition 2.4, which corresponds to the empty set (see Remark 6.2 (*iii*)).

We next give a character-theoretic description of the two parameter sets of interest.

Lemma 2.5. Let $n \ge 1$ and $r \ge 0$ be integers, and let $\epsilon \in \{1, -1\}$. Let G be an abelian group of order n^2 , and let D be an $r(n - \epsilon)$ -subset of G. Then D is a regular (n, r) Latin square type PDS in G (when $\epsilon = 1$), and is a regular (n, r) negative Latin square type PDS in G (when $\epsilon = -1$), if and only if

 $\chi(D) \in \{-\epsilon r, \epsilon(n-r)\}$ for all nonprincipal characters χ of G.

If D is a regular (n,r) Latin square type or negative Latin square type PDS in G, then the character set $\widetilde{D} = \{\chi \in \widehat{G} \mid \chi(D) = \epsilon(n-r)\}$ is also a regular (n,r) Latin square type or negative Latin square type PDS in \widehat{G} , respectively.

Proof. The first statement follows directly from Lemma 2.3. The second statement is obtained from [60, Thm. 3.4]: let D^+ be the dual of D as defined in [60], and note that $\tilde{D} = D^+$ in the case $\epsilon = 1$ and that $\tilde{D} = \hat{G} - 1_{\hat{G}} - D^+$ in the case $\epsilon = -1$.

We now briefly survey the numerous previous constructions of Latin square type and negative Latin square type partial difference sets, which we classify into three classes that are restricted to elementary abelian groups and five that are not. The three construction classes that are restricted to elementary abelian groups arise from the following sources.

(1) *Projective two-intersection sets.*

Projective two-intersection sets (sometimes called two-character sets) are classical configurations from finite geometry that provide a series of constructions [1, 2, 9, 22, 24, 25, 26, 27, 31, 47, 54, 73] (see also [60, Sect. 9]). Recently, projective two-intersection sets have been intensively studied in the context of intriguing sets in polar spaces (see [4, 5, 8, 12, 28, 29, 30, 42], for example). See [23] for a recent survey of intriguing sets, and [66, Sect. 4.5] for a detailed account of the connection between these sets and Latin square type and negative Latin square type PDSs. (2) Nondegenerate quadratic forms and their generalizations.

A fundamental construction employs nondegenerate quadratic forms over finite fields [60, Thm. 2.6]. This construction has been greatly extended to allow the quadratic form to be replaced by other functions, including bent and vectorial bent functions [14, 15, 17, 20, 44, 72, 86].

(3) Cyclotomic classes of a finite field.

A seminal construction due to Baumert, Mills and Ward involves uniform cyclotomy in a finite field [6], [60, Thm. 10.3]. Further cyclotomic constructions have been proposed, based on sophisticated number-theoretic techniques [62, 63, 65, 67, 68, 69, 71].

We now describe the five construction classes that are not restricted to elementary abelian groups. We shall later use LP-packings and NLP-packings to streamline many of these constructions, using only elementary methods. In some cases, we shall give a unified treatment for previously known parameter sets; in other cases, we shall produce examples with entirely new parameter sets.

(1) Direct constructions via partial congruence partitions

Definition 2.6. Let G be a group of order $n^2 > 1$. An (n, r) partial congruence partition of degree r in G is a collection $\{U_1, \ldots, U_r\}$ of order n subgroups of G such that $U_i \cap U_j = \{1_G\}$ for all $i \neq j$.

If U_1, \ldots, U_r is an (n, r) partial congruence partition in G, then $\sum_{i=1}^r (U_i - 1_G)$ is a regular (n, r) Latin square type partial difference set in G [3, Sect. 4.1]. A central objective in the study of (n, r) partial congruence partitions in groups of order n^2 is to determine the largest possible degree r [3, 55, 56]. The following result determines this value for abelian p-groups of the form $H \times H$.

Result 2.7 ([3, Thm. 2.7]). Let s_1, \ldots, s_v be nonnegative integers and let $m = \min\{s_i \mid s_i > 0\}$. Let p be prime and let $G = \prod_{i=1}^{v} \mathbb{Z}_{p^i}^{2s_i}$ and let $n = \sqrt{|G|}$. Then the largest integer r for which there exists an (n, r) partial congruence partition in G is $p^m + 1$. Therefore there exists a regular (n, r) Latin square type PDS in G for each positive integer $r \leq p^m + 1$.

We shall strengthen Result 2.7 using Corollary 5.4: see Remark 5.5 (i).

(2) Direct constructions via finite local rings

For positive integers p and w, define the function

$$\sigma(p,w) = \max_{\ell \mid w} \ell\left(\left\lfloor \frac{w/\ell - 2}{p} \right\rfloor + 1\right), \tag{2.2}$$

which determines the parameters of Latin square type partial difference sets in $\mathbb{Z}_{p^2}^w$ under the construction of [51, Corollary 3.4]. Various direct constructions of Latin square type partial difference sets exploit the structure of finite local rings [21, 38, 49, 50, 51, 52, 57, 59, 80, 84]. We summarize below the constructions from [21, 49, 50, 51, 52, 57, 59, 84], which we shall strengthen using Corollaries 5.4 and 5.7: see Remarks 5.5 (*ii*) and 5.8.

Result 2.8. Let p be prime. There exists an (n, r) Latin square type PDS in a group G of order n^2 for the following G and r, where $r \leq n$.

- (i) $G = \mathbb{Z}_{p^2}^w$ for w > 1, and r is a positive integer multiple of $\frac{n}{p^{\sigma(p,w)}}$ ([51, Thm. 3.2, Cor. 3.4]).
- (ii) $G = \mathbb{Z}_{p^2}^w \times \mathbb{Z}_p^{2uw}$ for w > 1 and a positive integer u, and r is a positive integer multiple of $\frac{n}{p^{\sigma(p,w)}}$ ([49, Thm. 3.1]).
- (iii) $G = \prod_{i=1}^{v} \mathbb{Z}_{p^{i}}^{2s_{i}}$ for nonnegative integers s_{1}, \ldots, s_{v} , and r is a positive integer multiple of $\frac{n}{p^{\operatorname{gcd}(s_{1},\ldots,s_{v})}}$ ([50, Thm. 3.4, Cor. 3.5]).
- (3) Lifting constructions via finite local rings

Finite local rings also provide constructions for Latin square type partial difference sets that lift examples from a group $G \times G$ to a larger group $G' \times G'$ [48, 50, 52]. We shall present in Theorem 4.4 a ring-free lifting construction that applies to a collection of Latin square type partial difference sets, rather than just a single one. This construction leads to the result stated in Theorem 1.1, which recovers the parameters of partial difference sets constructed via lifting in each of [48], [50, Sects. 4, 5], and [52].

(4) Product constructions

Various delicate product constructions for Latin square type and negative Latin square type partial difference sets have been found [62, 74, 75, 78, 79, 82]. The following examples occur in p-groups having arbitrarily large exponent.

Result 2.9 ([82, Cor. 6.2]). Let p be prime, and let u and s_2, s_4, \ldots, s_{2v} be nonnegative integers. Let $G = \mathbb{Z}_p^{2u} \times \prod_{i=1}^v \mathbb{Z}_{p^{2i}}^{2s_{2i}}$ and let $n = \sqrt{|G|}$. Then there is a partition of $G - 1_G$ into p regular partial difference sets in G, of which p - 1 are of $(n, \frac{n}{p})$ Latin square type and one is of $(n, \frac{n}{p} + 1)$ Latin square type.

We shall strengthen Result 2.9 using Corollary 5.4 (see Remark 5.5 (v)).

Result 2.10 ([82, Thm. 4.2, Cor. 5.1]).

- (i) Let u, w and s_4, s_6, \ldots, s_{2v} be nonnegative integers satisfying $u + w \ge 1$. Let $G = \mathbb{Z}_2^{4u} \times \mathbb{Z}_4^{2w} \times \prod_{i=2}^{v} \mathbb{Z}_{2^{2i}}^{4s_{2i}}$ and let $n = \sqrt{|G|}$. Then there is a partition of $G 1_G$ into 4 regular partial difference sets in G, of which three are of $(n, \frac{n}{4})$ negative Latin square type and one is of $(n, \frac{n}{4} 1)$ negative Latin square type.
- (ii) Let u and s_2, s_4, \ldots, s_{2v} be nonnegative integers. Let $G = \mathbb{Z}_3^{2u+2} \times \prod_{i=1}^v \mathbb{Z}_{3^{2i}}^{2s_{2i}}$ and let $n = \sqrt{|G|}$. Then there is a partition of $G 1_G$ into 3 regular partial difference sets in G, of which two are of $(n, \frac{n}{3})$ negative Latin square type and one is of $(n, \frac{n}{3} 1)$ negative Latin square type.

We shall strengthen Result 2.10 using Corollary 6.13 (see Remark 6.14).

(5) Constructions in Galois domains

A final construction class uses cyclotomy [46] and character sums [64, 70] over Galois domains (direct products of finite fields) to produce partial difference sets in the direct product of elementary abelian groups.

3. LP-Packings

In this section, we introduce a (c, t) LP-packing as a collection of t disjoint regular (tc, c) Latin square type PDSs, in an abelian group of order t^2c^2 , whose union satisfies an additional property. We provide examples of LP-packings, describe their relationship to partial congruence partitions, establish constraints on their structure, and present a product construction.

A principal motivation for this paper was the realization that in many previous constructions of such collections, possibly after some modification as summarized in Table 1.1, the *tc*-set $G - \sum_{i=1}^{t} P_i$ of elements avoided by the partial difference sets forms a subgroup U of G. Our definition of a (c, t) LP-packing is based on this observation.

Definition 3.1 (LP-packing). Let t > 1 and c > 0 be integers. Let G be an abelian group of order t^2c^2 , and let U be a subgroup of G of order tc. A (c,t) LP-packing in G relative to U is a collection $\{P_1, \ldots, P_t\}$ of t regular (tc, c) Latin square type PDSs in G for which $\sum_{i=1}^t P_i = G - U$.

Remark 3.2. In Definition 3.1, each P_i is a c(tc - 1)-subset of G, and |G - U| = tc(tc - 1), so the relation $\sum_{i=1}^{t} P_i = G - U$ is equivalent to the statement that the P_i are disjoint and their union is G - U.

We next give a characterization of a (c, t) LP-packing involving character sums.

Lemma 3.3. Let P_1, \ldots, P_t be c(tc-1)-subsets of an abelian group G of order t^2c^2 and let U be a subgroup of G of order tc. Then $\{P_1, \ldots, P_t\}$ is a (c, t) LP-packing in G relative to U if and only if the multiset equality

$$\{\chi(P_1), \dots, \chi(P_t)\} = \begin{cases} \{-c, \dots, -c\} & \text{if } \chi \in U^{\perp}, \\ \{(t-1)c, -c, \dots, -c\} & \text{if } \chi \notin U^{\perp} \end{cases}$$
(3.1)

holds for all nonprincipal characters χ of G.

Proof. Let χ be a nonprincipal character of G.

Suppose firstly that $\{P_1, \ldots, P_t\}$ is a (c, t) LP-packing in G relative to U. Apply χ to the equation $\sum_{i=1}^{t} P_i = G - U$ to give

$$\sum_{i=1}^t \chi(P_i) = \begin{cases} -tc & \text{if } \chi \in U^\perp, \\ 0 & \text{if } \chi \notin U^\perp. \end{cases}$$

Since $\chi(P_i) \in \{-c, (t-1)c\}$ for each *i* by Lemma 2.5, this implies that

$$\{\chi(P_1),\ldots,\chi(P_t)\} = \begin{cases} \{-c,\ldots,-c\} & \text{if } \chi \in U^{\perp}, \\ \{(t-1)c,-c,\ldots,-c\} & \text{if } \chi \notin U^{\perp}, \end{cases}$$

as required.

Now suppose that (3.1) holds. Then each P_i is a c(tc-1)-subset of G for which $\chi(P_i) \in \{-c, (t-1)c\}$, and so by Lemma 2.5 is a regular (tc, c) Latin square type PDS in G. It remains to show that $\sum_{i=1}^{t} P_i = G - U$. We have from $|P_i| = c(tc-1)$ and (3.1) that, for all $\chi \in \widehat{G}$,

$$\chi\Big(\sum_{i=1}^{t} P_i\Big) = \begin{cases} tc(tc-1) & \text{if } \chi \in G^{\perp}, \\ -tc & \text{if } \chi \in U^{\perp} \setminus G^{\perp}, \\ 0 & \text{if } \chi \notin U^{\perp}. \end{cases}$$

By the Fourier inversion formula (Proposition 2.2), this implies that $\sum_{i=1}^{t} P_i = G - U$, as required.

A (t, t+1) partial congruence partition (see Definition 2.6) lies at the heart of many configurations [40], [41, Sect. 4]. We next show that this structure can equivalently be expressed as a (1, t) LP-packing.

Proposition 3.4. Let G be an abelian group of order t^2 , let U_1, \ldots, U_t be t-subsets of G, and let U_0 be a subgroup of G of order t. Then $\{U_0, U_1, \ldots, U_t\}$ is a (t, t + 1) partial congruence partition in G if and only if $\{U_1 - 1_G, \ldots, U_t - 1_G\}$ is a (1, t) LP-packing in G relative to U_0 .

Proof. By Definition 2.6, $\{U_0, U_1, \ldots, U_t\}$ is a (t, t + 1) partial congruence partition in G if and only if each of U_1, \ldots, U_t is a subgroup of G and $\sum_{i=0}^t U_i = G + t1_G$. By Definitions 2.4 and 3.1, $\{U_1 - 1_G, \ldots, U_t - 1_G\}$ is a (1, t) LP-packing in G relative to U_0 if and only if each of $U_1 - 1_G, \ldots, U_t - 1_G$ is a regular $(t^2, t - 1, t - 2, 0)$ PDS in G and $\sum_{i=1}^t (U_i - 1_G) = G - U_0$. It is therefore sufficient to show for $i = 1, \ldots, t$ that U_i is a subgroup of G if and only if $U_i - 1_G$ is a regular $(t^2, t - 1, t - 2, 0)$ partial difference set in G. This follows directly from Definition 2.1.

The following result greatly constrains the degree of a partial congruence partition.

Result 3.5. Suppose there exists an (n, r) partial congruence partition of degree r in an abelian group G of order n^2 . Then $r \leq n + 1$, and

- (i) if $r \ge \lfloor \sqrt{n} \rfloor + 2$ then G is elementary abelian ([55, Thms. 3.1, 3.4]).
- (ii) if $r = \sqrt{n} + 1$ then either G is elementary abelian or $G = (\mathbb{Z}_{p^2})^w$ for some prime p and positive integer w ([56, Cor. 5.3]).

By Result 3.5 (i), a (t, t + 1) partial congruence partition in an abelian group G can exist only when $t = p^s$ for some prime p and positive integer s, and $G \cong \mathbb{Z}_p^{2s}$. By Proposition 3.4, such a structure exists if and only if there is a $(1, p^s)$ LP-packing in G relative to an order p^s subgroup. We now show, in an example we shall refer to frequently in the rest of the paper, that this structure exists for all primes p and positive integers s by reference to a spread of a vector space over a finite field. **Definition 3.6.** Let p be prime and s a positive integer, and let V be a 2s-dimensional vector space over the field \mathbb{F}_p with identity 1_V . A *spread* of V is a $(p^s + 1)$ -set $\{H_0, H_1, \ldots, H_{p^s}\}$ of s-dimensional vector subspaces of V such that $H_i \cap H_j = \{1_V\}$ for all $i \neq j$.

Example 3.7. Let p be prime and s a positive integer, and let V be a 2s-dimensional vector space over the field \mathbb{F}_p with identity 1_V . Then there exists a spread $\{H_0, H_1, \ldots, H_{p^s}\}$ of V [53, Chap. VII, Sect. 5]. By Definitions 2.6 and 3.6, $\{H_0, \ldots, H_{p^s}\}$ forms a $(p^s, p^s + 1)$ partial congruence partition in V, because H_i is a subgroup of order p^s in $V \cong \mathbb{Z}_p^{2s}$ if and only if H_i is an s-dimensional vector subspace of V. Then by Proposition 3.4 with $t = p^s$, we see that $\{H_1 - 1_V, \ldots, H_{p^s} - 1_V\}$ is a $(1, p^s)$ LP-packing in $V \cong \mathbb{Z}_p^{2s}$ relative to $H_0 \cong \mathbb{Z}_p^s$, and by Lemma 3.3 each nonprincipal character of V is principal on exactly one of H_0, \ldots, H_{p^s} .

Remark 3.8. According to Definition 3.1, a (c, t) LP-packing involves a collection of t disjoint regular (tc, c) Latin square type PDSs in an abelian group G of order t^2c^2 . Since this collection covers all but tc elements of G, and each PDS contains c(tc - 1) elements, the collection has maximum possible size when c > 1. The collection is not necessarily of maximum possible size when c = 1: for p prime and $t = p^s$, the (t, t+1) partial congruence partition of Example 3.7 is a collection of t + 1 disjoint regular (t, 1) Latin square type PDSs in the elementary abelian group of order t^2 . However, as Result 3.5 shows, if we instead attempt to create a partial congruence partition in a non-elementary abelian group of order t^2 then the degree must drop from t +1 to at most $\lfloor \sqrt{t} \rfloor + 1$, covering a proportion of only about $1/\sqrt{t}$ of the elements of G. In contrast, in Section 5 we shall provide constructions of (c, t) LP-packings with c > 1 in various nonelementary abelian groups. For this reason, we regard a (c, t) LP-packing with c > 1 as a natural generalization of a (t, t + 1) partial congruence partition.

The characterization in Lemma 3.3, together with Lemma 2.5, shows that we can combine subsets from an LP-packing to obtain Latin square type PDSs with various parameters.

Lemma 3.9. Suppose that $\{P_1, \ldots, P_t\}$ is a (c, t) LP-packing in an abelian group G of order t^2c^2 relative to a subgroup U of order tc. Let I be a b-subset of $\{1, \ldots, t\}$. Then

- (i) $\sum_{i \in I} P_i$ is a regular (tc, bc) Latin square type PDS in G.
- (*ii*) $\sum_{i \in I} P_i + U 1_G$ is a regular (tc, bc + 1) Latin square type PDS in G.

Remark 3.10. Suppose that $\{P_1, \ldots, P_t\}$ is a (c, t) LP-packing in an abelian group G of order t^2c^2 relative to a subgroup U of order tc.

- (i) By Definition 3.1 and Lemma 3.9, we see that $\{P_1, \ldots, P_{t-1}, P_t + U 1_G\}$ is a partition of $G 1_G$ into t regular partial difference sets in G, of which the first t 1 are of (tc, c) Latin square type and the last is of (tc, c + 1) Latin square type.
- (ii) For each *i*, let $\widetilde{P_i} = \{\chi \in \widehat{G} \mid \chi(P_i) = (t-1)c\}$. By Lemma 2.5, each $\widetilde{P_i}$ is a regular (tc, c) Latin square type PDS in \widehat{G} , and by Lemma 3.3 each $\chi \in \widehat{G} \setminus U^{\perp}$ belongs to exactly one $\widetilde{P_i}$ and each $\chi \in U^{\perp}$ belongs to no $\widetilde{P_i}$. By Definition 3.1, the collection $\{\widetilde{P_1}, \ldots, \widetilde{P_t}\}$ is therefore a (c, t) LP-packing in \widehat{G} relative to U^{\perp} .

Combining the subsets of an LP-packing into equally-sized collections gives an LP-packing with fewer subsets.

Lemma 3.11. Suppose there exists a (c,t) LP-packing in an abelian group G of order t^2c^2 relative to a subgroup U of order tc, and suppose s divides t. Then there exists an $(sc, \frac{t}{s})$ LP-packing in G relative to U.

Proof. Let $\{P_1, \ldots, P_t\}$ be a (c, t) LP-packing in G relative to U, and let

$$P'_i = P_{is+1} + P_{is+2} + \dots + P_{is+s}$$
 for $i = 0, 1, \dots, \frac{t}{s} - 1$.

Then Lemma 3.3 shows that $\{P'_0, \ldots, P'_{\frac{t}{s}-1}\}$ is an $(sc, \frac{t}{s})$ LP-packing in G relative to U.

The following result, which is inspired by [36, Lemma 3.2], allows us to establish some constraints on LP-packings in Proposition 3.13. These constraints assist in finding examples computationally, as demonstrated in Example 3.14.

Lemma 3.12. Let G be a group of order n^2 and let U be a subgroup of order n. Let $\{g_0, g_1, \ldots, g_{n-1}\}$ be a complete set of right coset representatives for U in G, where $Ug_0 = U$. Suppose P is a regular (n, r) Latin square type PDS in G for which $P \cap U = \emptyset$. Then $|P \cap Ug_i| = r$ for each $i = 1, \ldots, n-1$.

Proof. Since $P \cap U = \emptyset$,

$$r(n-1) = |P| = \sum_{i=1}^{n-1} |P \cap Ug_i|.$$
(3.2)

For all $x, y \in P$, we have $xy^{-1} \in U$ if and only if x, y belong to the same right coset of U. Therefore

$$\sum_{i=1}^{n-1} |P \cap Ug_i|^2 = \sum_{g \in U} [PP^{(-1)}]_g.$$
(3.3)

Now P is an $(n^2, r(n-1), n + r^2 - 3r, r^2 - r)$ PDS in G, and so by (2.1)

$$PP^{(-1)} = r(n-r) + (n-2r)P + (r^2 - r)G.$$

Since $P \cap U = \emptyset$, this implies that

$$\sum_{g \in U} [PP^{(-1)}]_g = r(n-r) + (r^2 - r)n = r^2(n-1).$$

Substitution in (3.3) then gives

$$\sum_{i=1}^{n-1} |P \cap Ug_i|^2 = r^2(n-1),$$

which, by the Cauchy-Schwarz inequality and (3.2), gives $|P \cap Ug_i| = r$ for each i = 1, ..., n-1 as required.

Combinatorial theory 27 (2020), #P00

Proposition 3.13. Let G be an abelian group of order t^2c^2 , and let U be a subgroup of G of order tc. Let $\{g_0, g_1, \ldots, g_{tc-1}\}$ be a complete set of right coset representatives for U in G, where $Ug_0 = U$, and define subsets S, N of $\{1, \ldots, tc - 1\}$ by

$$S = \{j \mid Ug_j = Ug_j^{-1}\},\$$

N contains exactly one of j, k when $Ug_j \neq Ug_j^{-1} = Ug_k$.

Suppose that $\{P_1, \ldots, P_t\}$ is a (c, t) LP-packing in G relative to U. Then there are c-subsets H_{ij} of U for $i = 1, \ldots, t$ and $j = 1, \ldots, tc - 1$ satisfying

$$\begin{split} P_{i} &= \sum_{j \in S} H_{ij}g_{j} + \sum_{j \in N} \left(H_{ij}g_{j} + H_{ij}^{(-1)}g_{j}^{-1} \right) \quad \textit{for each } i, \\ H_{ij}^{(-1)}g_{j}^{-1} &= H_{ij}g_{j} \quad \textit{for } j \in S, \\ H_{ij}^{(-1)}g_{j}^{-1} &= H_{ik}g_{k} \quad \textit{for } j \in N \textit{ where } Ug_{j}^{-1} = Ug_{k}, \\ U &= \sum_{i=1}^{t} H_{ij} \quad \textit{for each } j. \end{split}$$

Proof. By Definition 3.1, each P_i is a regular (tc, c) Latin square type PDS in G, and G - U is the disjoint union of the P_i . Write

$$P_{i} = \sum_{j=1}^{tc-1} (P_{i} \cap Ug_{j})$$

= $\sum_{j \in S} (P_{i} \cap Ug_{j}) + \sum_{j \in N} ((P_{i} \cap Ug_{j}) + (P_{i} \cap Ug_{j}^{-1})).$ (3.4)

By Lemma 3.12, for each i = 1, ..., t and each j = 1, ..., tc - 1 we may write

 $P_i \cap Ug_j = H_{ij}g_j$ for some *c*-subset H_{ij} of *U*.

Substitute in (3.4) and use $P_i = P_i^{(-1)}$ to give

$$P_i = \sum_{j \in S} H_{ij}g_j + \sum_{j \in N} \left(H_{ij}g_j + H_{ij}^{(-1)}g_j^{-1} \right),$$

and the constraints $H_{ij}^{(-1)}g_j^{-1} = H_{ij}g_j$ for $j \in S$, and $H_{ij}^{(-1)}g_j^{-1} = H_{ik}g_k$ for $j \in N$ where $Ug_j^{-1} = Ug_k$. For each j, we have $Ug_j = \sum_{i=1}^t (P_i \cap Ug_j) = \sum_{i=1}^t H_{ij}g_j$ and therefore $U = \sum_{i=1}^t H_{ij}$.

Example 3.14. Using Proposition 3.13, we obtain the following exhaustive search results for LP-packings in \mathbb{Z}_4^3 relative to two nonisomorphic order 8 subgroups.

(i) There are exactly 1536 distinct (4, 2) LP-packings $\{P_1, P_2\}$ in $\mathbb{Z}_4^3 = \langle x, y, z \rangle$ relative to $(2\mathbb{Z}_4) \times (2\mathbb{Z}_4) \times (2\mathbb{Z}_4) = \langle x^2, y^2, z^2 \rangle$, one of which is

$$\begin{split} P_1 = & (1 + x^2 + y^2 + x^2y^2)x + (x^2 + x^2y^2 + z^2 + y^2z^2)y + (x^2 + x^2z^2 + y^2 + y^2z^2)z \\ & + (y^2 + z^2 + x^2 + x^2y^2z^2)xy + (1 + x^2 + z^2 + x^2z^2)xz \\ & + (x^2 + x^2y^2 + x^2z^2 + x^2y^2z^2)yz + (y^2 + z^2 + x^2y^2 + x^2z^2)xyz, \\ P_2 = & \langle x, y, z \rangle - \langle x^2, y^2, z^2 \rangle - P_1. \end{split}$$

The presented form of P_1 shows that both P_1 and P_2 intersect the seven nonidentity cosets of the subgroup $\langle x^2, y^2, z^2 \rangle$ in $\langle x, y, z \rangle$ in exactly 4 elements.

(*ii*) There are exactly 512 distinct (4, 2) LP-packings $\{R_1, R_2\}$ in $\mathbb{Z}_4^3 = \langle x, y, z \rangle$ relative to $\mathbb{Z}_4 \times (2\mathbb{Z}_4) = \langle x, y^2 \rangle$, one of which is

$$\begin{split} R_1 = & (y^2 + xy^2 + x^2y^2 + x^3)z + (x + xy^2 + x^3 + x^3y^2)z^2 + (y^2 + x + x^2y^2 + x^3y^2)z^3 \\ & + (x + x^2 + x^2y^2 + x^3y^2)y + (1 + x^2 + x^3 + x^3y^2)yz \\ & + (x + x^2 + x^2y^2 + x^3y^2)yz^2 + (1 + x + xy^2 + x^2)y^3z^3, \\ R_2 = & \langle x, y, z \rangle - \langle x, y^2 \rangle - R_1. \end{split}$$

The presented form of R_1 shows that both R_1 and R_2 intersect the seven nonidentity cosets of the subgroup $\langle x, y^2 \rangle$ in $\langle x, y, z \rangle$ in exactly 4 elements.

We conclude this section with a product construction for LP-packings.

Theorem 3.15. For j = 1, 2, suppose there exists a (c_j, t) LP-packing in an abelian group G_j of order $t^2c_j^2$ relative to a subgroup U_j of order tc_j . Then there exists a (tc_1c_2, t) LP-packing in $G_1 \times G_2$ relative to $U_1 \times U_2$.

Proof. For j = 1, 2, let $\{P_{j,0}, \ldots, P_{j,t-1}\}$ be a (c_j, t) LP-packing in G_j relative to U_j and define

$$K_{\ell} = P_{1,\ell}U_2 + U_1P_{2,\ell} + \sum_{i=0}^{t-1} P_{1,i}P_{2,i+\ell}$$
 for $\ell = 0, \dots, t-1$,

where the subscript $i+\ell$ is reduced modulo t. We shall use Lemma 3.3 to show that $\{K_0, K_1, \ldots, K_{t-1}\}$ is a (tc_1c_2, t) LP-packing in $G_1 \times G_2$ relative to $U_1 \times U_2$.

Each $P_{j,i}$ is a $c_j(tc_j - 1)$ -subset of $G_j - U_j$, so each K_ℓ is a subset of $G_1 \times G_2$ of size $c_1(tc_1 - 1)tc_2 + tc_1c_2(tc_2 - 1) + \sum_{i=0}^{t-1} c_1(tc_1 - 1)c_2(tc_2 - 1) = tc_1c_2(t^2c_1c_2 - 1)$. Let χ be a nonprincipal character of $G_1 \times G_2$, and let $\chi_j = \chi|_{G_j}$ for j = 1, 2. Then

$$\chi(K_{\ell}) = \chi_1(P_{1,\ell})\chi_2(U_2) + \chi_1(U_1)\chi_2(P_{2,\ell}) + \sum_{i=0}^{t-1}\chi_1(P_{1,i})\chi_2(P_{2,i+\ell}) \quad \text{for } \ell = 0, \dots, t-1.$$

By Lemma 3.3, it remains to prove that

$$\{\chi(K_0), \dots, \chi(K_{t-1})\} = \begin{cases} \{-tc_1c_2, \dots, -tc_1c_2\} & \text{if } \chi \in (U_1 \times U_2)^{\perp}, \\ \{t(t-1)c_1c_2, -tc_1c_2, \dots, -tc_1c_2\} & \text{if } \chi \notin (U_1 \times U_2)^{\perp}. \end{cases}$$
(3.5)

For j = 1, 2, by Lemma 3.3 we are given that

$$\{\chi_j(P_{j,0}), \dots, \chi_j(P_{j,t-1})\} = \begin{cases} \{c_j(tc_j-1), \dots, c_j(tc_j-1)\} & \text{if } \chi_j \in G_j^{\perp}, \\ \{-c_j, \dots, -c_j\} & \text{if } \chi_j \in U_j^{\perp} \setminus G_j^{\perp}, \\ \{(t-1)c_j, -c_j, \dots, -c_j\} & \text{if } \chi_j \notin U_j^{\perp}. \end{cases}$$

Since χ is nonprincipal on $G_1 \times G_2$, we may assume by symmetry that χ_2 is nonprincipal on G_2 . Again by symmetry, we need to consider only the following three cases, which together establish (3.5).

Case 1: χ_1 is principal on U_1 and χ_2 is principal on U_2 .

Then $\chi_1(P_{1,i})$ is constant over *i* (regardless of whether χ_1 is principal or nonprincipal on G_1), so $\chi(K_\ell) = \chi_1(P_{1,\ell})tc_2 + tc_1(-c_2) + \chi_1(P_{1,\ell})\sum_{i=0}^{t-1}(-c_2) = -tc_1c_2$ for each ℓ .

- **Case 2:** χ_1 is principal on U_1 and χ_2 is nonprincipal on U_2 . Then $\chi_1(P_{1,i})$ is again constant over i and $\sum_{i=0}^{t-1} \chi_2(P_{2,i}) = 0$, so $\chi(K_\ell) = tc_1\chi_2(P_{2,\ell})$ which gives $\{\chi(K_0), \ldots, \chi(K_{t-1})\} = \{t(t-1)c_1c_2, -tc_1c_2, \ldots, -tc_1c_2\}.$
- **Case 3:** χ_1 is nonprincipal on U_1 and χ_2 is nonprincipal on U_2 . Then $\chi(K_\ell) = \sum_{i=0}^{t-1} \chi_1(P_{1,i})\chi_2(P_{2,i+\ell})$, so the multiset $\{\chi(K_0), \dots, \chi(K_{t-1})\}$ contains one occurrence of the value $(t-1)c_1(t-1)c_2 + (t-1)(-c_1)(-c_2) = t(t-1)c_1c_2$, and t-1 occurrences of the value $(t-1)c_1(-c_2) + (-c_1)(t-1)c_2 + (t-2)(-c_1)(-c_2) = -tc_1c_2$.

Remark 3.16. The construction of Theorem 3.15 is closely modelled on the product construction for collections of Latin square type PDSs presented by Polhill in [82, Thm. 2.2]. Indeed, the configuration described in [82, Lemma 2.1] for e = 1 can be represented as $\{P_1, \ldots, P_{t-1}, P_t + U - 1_G\}$, where $\{P_1, \ldots, P_t\}$ is a (c, t) LP-packing in G relative to U (see also Remark 3.10 (i)). Our construction has the advantage that, by identifying the role of the avoided subgroup U in the definition of an LP-packing, we are able to control the avoided subgroup $U_1 \times U_2$ in Theorem 3.15.

We extend the construction of Theorem 3.15 using Lemma 3.11.

Corollary 3.17. For j = 1, 2, suppose there exists a (c_j, t_j) LP-packing in an abelian group G_j of order $t_j^2 c_j^2$ relative to a subgroup U_j of order $t_j c_j$, and suppose that t_1 divides t_2 . Then there exists a $(t_2c_1c_2, t_1)$ LP-packing in $G_1 \times G_2$ relative to $U_1 \times U_2$.

Proof. By Lemma 3.11 with $s = \frac{t_2}{t_1}$, there exists a $(\frac{t_2}{t_1}c_2, t_1)$ LP-packing in G_2 relative to U_2 . Then by Theorem 3.15, there exists a $(t_2c_1c_2, t_1)$ LP-packing in $G_1 \times G_2$ relative to $U_1 \times U_2$. \Box

4. LP-partitions

In this section, we introduce a (c, t) LP-partition as an auxiliary configuration in the construction of an LP-packing in a larger group from an LP-packing in a smaller group. We then show how

to construct an LP-partition from a collection of LP-partitions in various factor groups. We shall apply these two constructions recursively in Section 5 to produce infinite families of LP-packings.

Definition 4.1 (LP-partition). Let t > 1 and c > 0 be integers. Let G be an abelian group of order t^2c^2 , let V be a subgroup of G of order tc^2 , and let $H \leq V$. A (c, t) LP-partition in G - V relative to H is a collection $\{R_1, \ldots, R_t\}$ of t disjoint $(t - 1)c^2$ -subsets of G whose union is G - V and for which the multiset equality

$$\{\chi(R_1), \dots, \chi(R_t)\} = \begin{cases} \{-c^2, \dots, -c^2\} & \text{if } \chi \in V^{\perp}, \\ \{0, \dots, 0\} & \text{if } \chi \in H^{\perp} \setminus V^{\perp}, \\ \{(t-1)c, -c, \dots, -c\} & \text{if } \chi \notin H^{\perp}. \end{cases}$$

holds for all nonprincipal characters χ of G.

Remark 4.2.

(i) In Definition 4.1, we can deduce that the R_i are disjoint and their union is G - V from the other conditions, by applying the Fourier inversion formula to the equation

$$\chi\Big(\sum_{i=1}^{t} R_i\Big) = \begin{cases} t(t-1)c^2 & \text{if } \chi \in G^{\perp}, \\ -tc^2 & \text{if } \chi \in V^{\perp} \setminus G^{\perp}, \\ 0 & \text{if } \chi \notin V^{\perp}. \end{cases}$$

(*ii*) Each subset of a (c, t) LP-partition in G - V relative to H is a (G, V, H; c) Latin shell, as introduced in [52, Defn. 3.1] as an auxiliary configuration in the construction of a single Latin square type PDS.

A simple example of an LP-partition is given by a spread of a vector space (see Definition 3.6).

Example 4.3. By Lemma 3.3 and Definition 4.1, a (1, t) LP-packing in G relative to U is identical to a (1, t) LP-partition in G - U relative to U. Let $t = p^s$ for a prime p and positive integer s, and let V be a 2s-dimensional vector space over \mathbb{F}_p with identity 1_V . Let $\{H_0, \ldots, H_{p^s}\}$ be a spread of $V \cong \mathbb{Z}_p^{2s}$. Following Example 3.7, $\{H_1 - 1_V, \ldots, H_{p^s} - 1_V\}$ is therefore a $(1, p^s)$ LP-partition in $V - H_0$ relative to H_0 as well as a $(1, p^s)$ LP-packing in V relative to H_0 .

We next combine the subsets of an LP-partition with the subsets obtained by lifting an LP-packing, in order to produce an LP-packing in a larger group.

Theorem 4.4. Let G be an abelian group of order t^4c^2 containing subgroups $H \leq U \leq V \leq G$, where H, U, V have order t, t^2c , t^3c^2 , respectively. Suppose there exists a (tc,t) LP-partition in G - V relative to H, and a (c,t) LP-packing in V/H relative to U/H. Then there exists a (tc,t) LP-packing in G relative to U.

Combinatorial theory 27 (2020), #P00

Proof. Let $\{R_1, \ldots, R_t\}$ be a (tc, t) LP-partition in G - V relative to H, and let $\{P_1, \ldots, P_t\}$ be a (c, t) LP-packing in V/H relative to U/H. For $i = 1, \ldots, t$, let $P'_i = \{g \in V \mid gH \in P_i\}$ be the pre-image of P_i under the quotient mapping from V to V/H. We shall use Lemma 3.3 to show that $\{P'_1 + R_1, \ldots, P'_t + R_t\}$ is a (tc, t) LP-packing in G relative to U.

Each P_i is a c(tc-1)-subset of V/H - U/H by Definition 3.1, so each P'_i is a tc(tc-1)-subset of V - U. Since each R_i is a $(t-1)(tc)^2$ -subset of G - V by Definition 4.1, it follows that each $P'_i + R_i$ is a subset of G of size $|P'_i| + |R_i| = tc(t^2c-1)$.

By Lemma 3.3, for all characters ψ of V/H we have

$$\{\psi(P_1), \dots, \psi(P_t)\} = \begin{cases} \{c(tc-1), \dots, c(tc-1)\} & \text{if } \psi \in (V/H)^{\perp}, \\ \{-c, \dots, -c\} & \text{if } \psi \in (U/H)^{\perp} \setminus (V/H)^{\perp}, \\ \{(t-1)c, -c, \dots, -c\} & \text{if } \psi \notin (U/H)^{\perp}. \end{cases}$$

Let χ be a nonprincipal character of G. If χ is principal on H then it induces a nonprincipal character ψ on G/H. Since P'_i is a union of cosets of the order t subgroup H, we obtain

$$\chi(P'_i) = \begin{cases} t\psi(P_i) & \text{if } \chi \in H^{\perp}, \\ 0 & \text{if } \chi \notin H^{\perp}. \end{cases}$$

Therefore the value of $\{\chi(P'_1), \ldots, \chi(P'_t)\}$ is as specified in the second column of the following table:

	$\{\chi(P'_1),\ldots,\chi(P'_t)\}$	$\{\chi(R_1),\ldots,\chi(R_t)\}$	$\{\chi(P'_1+R_1),\ldots,\chi(P'_t+R_t)\}$
$\chi \in V^\perp \setminus G^\perp$	$\{tc(tc-1),\ldots,tc(tc-1)\}$	$\{-t^2c^2,\ldots,-t^2c^2\}$	$\left\{ -tc -tc \right\}$
$\chi \in U^\perp \setminus V^\perp$	$\{-tc,\ldots,-tc\}$	رم 0}	$1-\iota c, \ldots, -\iota c$
$\chi \in H^{\perp} \setminus U^{\perp}$	$\{t(t-1)c, -tc, \dots, -tc\}$	$10, \ldots, 0f$	$\int t(t-1)c - tc - tc$
$\chi \notin H^\perp$	$\{0,\ldots,0\}$	$\{t(t-1)c, -tc, \dots, -tc\}$	$\iota(\iota-1)\iota, -\iota\iota, \ldots, -\iota\iota)$

By Definition 4.1, the value of $\{\chi(R_1), \ldots, \chi(R_t)\}$ is as given in the third column of the table, and the fourth column contains the sum of the second and third columns. Lemma 3.3 then implies that $\{P'_1 + R_1, \ldots, P'_t + R_t\}$ is a (tc, t) LP-packing in G relative to U, as required.

Remark 4.5. Lifting constructions similar to that of Theorem 4.4 were proposed in [48], [50, Sects. 4, 5], [52], each producing a single Latin square type partial difference set. These previous constructions make use of the delicate structure of finite local rings. In contrast, the construction of Theorem 4.4 applies simultaneously to a collection of such partial difference sets, and does not require ring theory.

We now construct an LP-partition by lifting and combining the subsets of a collection of LP-partitions in various factor groups. The construction makes use of two spreads of a vector space over a finite field (as introduced in Definition 3.6).

Theorem 4.6. Let $t = p^s$ for a prime p and positive integer s. Let G be an abelian group of order t^4c^2 containing subgroups $Q \leq G' \leq G$, where $Q \cong \mathbb{Z}_p^{2s}$ and $G/G' \cong \mathbb{Z}_p^{2s}$. Let H_0, \ldots, H_t be subgroups of G forming a spread when viewed as subgroups of Q. Let V_0, \ldots, V_t be subgroups of G for which $\{V_0/G', \ldots, V_t/G'\}$ is a spread of G/G', and for which $H_i \leq V_i$ for each $i = 1, \ldots, t$. Suppose that for each $i = 1, \ldots, t$ there exists a (c, t) LP-partition in $V_i/H_i - G'/H_i$ relative to Q/H_i . Then there exists a (tc, t) LP-partition in $G - V_0$ relative to H_0 .

Proof. For each i = 1, ..., t, let $\{S_{i1}, ..., S_{it}\}$ be a (c, t) LP-partition in $V_i/H_i - G'/H_i$ relative to Q/H_i . By Definition 4.1, for each i = 1, ..., t and for all nonprincipal characters ψ_i of V_i/H_i we have

$$\{\psi_i(S_{i1}), \dots, \psi_i(S_{it})\} = \begin{cases} \{-c^2, \dots, -c^2\} & \text{if } \psi_i \in (G'/H_i)^{\perp}, \\ \{0, \dots, 0\} & \text{if } \psi_i \in (Q/H_i)^{\perp} \setminus (G'/H_i)^{\perp}, \\ \{(t-1)c, -c, \dots, -c\} & \text{if } \psi_i \notin (Q/H_i)^{\perp}. \end{cases}$$

For each i, j satisfying $1 \leq i, j \leq t$, let $S'_{ij} = \{g \in V_i \mid gH_i \in S_{ij}\}$ be the pre-image of S_{ij} under the quotient mapping from V_i to V_i/H_i , and let $R_j = \sum_{i=1}^t S'_{ij}$ for $j = 1, \ldots, t$. We shall show that $\{R_1, \ldots, R_t\}$ is a (tc, t) LP-partition in $G - V_0$ relative to H_0 .

By Definition 4.1, each S_{ij} is a $(t-1)c^2$ -subset of $V_i/H_i - G'/H_i$, and so each S'_{ij} is a $t(t-1)c^2$ -subset of $V_i - G'$. By Example 3.7, the $V_i/G' - 1_{G/G'}$ are disjoint in G/G' and so the $V_i - G'$ are disjoint in G. Therefore each $R_j = \sum_{i=1}^t S'_{ij}$ is a $t^2(t-1)c^2$ -subset of G. By Definition 4.1 and Remark 4.2 (i), it is now sufficient to show that, for all nonprincipal characters χ of G,

$$\{\chi(R_1), \dots, \chi(R_t)\} = \begin{cases} \{-t^2 c^2, \dots, -t^2 c^2\} & \text{if } \chi \in V_0^{\perp}, \\ \{0, \dots, 0\} & \text{if } \chi \in H_0^{\perp} \setminus V_0^{\perp}, \\ \{t(t-1)c, -tc, \dots, -tc\} & \text{if } \chi \notin H_0^{\perp}. \end{cases}$$
(4.1)

Let χ be a nonprincipal character of G. Since the H_i correspond to a spread of Q, if χ is nonprincipal on Q, then χ is principal on exactly one of the H_i (see Example 3.7). If χ is nonprincipal on V_i and principal on H_i , then χ induces a nonprincipal character ψ_i on V_i/H_i ; since each S'_{ij} is a union of cosets of the order t subgroup H_i , we have for each i, j satisfying $1 \leq i, j \leq t$ that

$$\chi(S'_{ij}) = \begin{cases} t\psi_i(S_{ij}) & \text{if } \chi \in H_i^{\perp} \setminus V_i^{\perp}, \\ 0 & \text{if } \chi \notin H_i^{\perp}. \end{cases}$$

If χ is principal on G' then it induces a nonprincipal character τ on G/G'; since the V_i/G' form a spread of G/G', we then have that τ is principal on exactly one of the V_i/G' and so χ is principal on exactly one of the V_i . We shall use the subgroup inclusions

$$H_i \leqslant Q \leqslant G' \leqslant V_i \quad \text{for each } i = 0, \dots, t.$$
 (4.2)

The conclusions of the following five cases collectively establish (4.1).

Combinatorial theory **27** (2020), #P00

Case 1: χ is principal on V_0 .

Then by (4.2), χ is principal on G' and on each H_i . For each $i = 1, \ldots, t$, therefore χ is nonprincipal on V_i and principal on G'/H_i . For each j, we obtain $\chi(R_j) = \sum_{i=1}^t \chi(S'_{ij}) = t \sum_{i=1}^t \psi_i(S_{ij}) = t \sum_{i=1}^t (-c^2) = -t^2c^2$.

Case 2: χ is principal on G' and nonprincipal on V_0 .

Then by (4.2), χ is principal on each H_i . Therefore χ is principal on each G'/H_i and on exactly one of V_1, \ldots, V_t , say V_I . For each j, we obtain $\chi(R_j) = \sum_{i=1}^t \chi(S'_{ij}) =$ $|S'_{Ij}| + t \sum_{1 \le i \le t, i \ne I} \psi_i(S_{ij}) = t(t-1)c^2 + t \sum_{1 \le i \le t, i \ne I} (-c^2) = 0.$

Case 3: χ is principal on Q and nonprincipal on G'. Then by (4.2), χ is principal on each H_i and nonprincipal on each V_i . Therefore χ is principal on each Q/H_i and nonprincipal on each G'/H_i . For each i we obtain $\chi(B_i) =$

principal on each Q/H_i and nonprincipal on each G'/H_i . For each j, we obtain $\chi(R_j) = t \sum_{i=1}^t \psi_i(S_{ij}) = 0$.

Case 4: χ is principal on H_0 and nonprincipal on Q.

Then for i = 1, ..., t, we have that χ is nonprincipal on H_i and so $\chi(S'_{ij}) = 0$ for each j. Therefore $\chi(R_i) = 0$ for each j.

Case 5: χ is nonprincipal on H_0 .

By (4.2), χ is nonprincipal on Q and on each V_i . Therefore χ is principal on exactly one of H_1, \ldots, H_t , say H_I , and is nonprincipal on Q/H_I . For each j, we obtain $\chi(R_j) = t\psi_I(S_{Ij})$ and so $\{\chi(R_1), \ldots, \chi(R_t)\} = \{t(t-1)c, -tc, \ldots, -tc\}.$

The following example illustrates the use of Theorems 4.4 and 4.6 to construct a (4, 4) LPpacking in \mathbb{Z}_4^4 relative to each of the order 16 subgroups $\mathbb{Z}_4 \times (2\mathbb{Z}_4)^2$ and \mathbb{Z}_4^2 and $(2\mathbb{Z}_4)^4$ in turn, starting from a spread of \mathbb{F}_4^2 . In Section 5 we shall show how to apply this procedure recursively to produce infinite families of LP-packings and LP-partitions in groups of increasing exponent and fixed rank.

Example 4.7. Let $(\mathbb{F}_4, \oplus, \cdot) = \{0, 1, \alpha, \alpha^2\}$, where $\alpha^2 = \alpha \oplus 1$. Then a spread of (\mathbb{F}_4^2, \oplus) is $\{\langle (1,0) \rangle, \langle (0,1) \rangle, \langle (1,1) \rangle, \langle (\alpha,1) \rangle, \langle (\alpha \oplus 1,1) \rangle \}$. Using the isomorphism from (\mathbb{F}_4^2, \oplus) to $\mathbb{Z}_2^4 = \langle y_1, y_2, y_3, y_4 \rangle$ given by $(1,0) \mapsto y_1, (\alpha,0) \mapsto y_2, (0,1) \mapsto y_3, (0,\alpha) \mapsto y_4$, this spread in multiplicative notation is

$$\left\{ \langle y_1, y_2 \rangle, \langle y_3, y_4 \rangle, \langle y_1 y_3, y_2 y_4 \rangle, \langle y_2 y_3, y_1 y_2 y_4 \rangle, \langle y_1 y_2 y_3, y_1 y_4 \rangle \right\}.$$

$$(4.3)$$

Let

$$S_{1}(y_{1}, y_{2}, y_{3}, y_{4}) = y_{3} + y_{4} + y_{3}y_{4},$$

$$S_{2}(y_{1}, y_{2}, y_{3}, y_{4}) = y_{1}y_{3} + y_{2}y_{4} + y_{1}y_{2}y_{3}y_{4},$$

$$S_{3}(y_{1}, y_{2}, y_{3}, y_{4}) = y_{2}y_{3} + y_{1}y_{2}y_{4} + y_{1}y_{3}y_{4},$$

$$S_{4}(y_{1}, y_{2}, y_{3}, y_{4}) = y_{1}y_{2}y_{3} + y_{1}y_{4} + y_{2}y_{3}y_{4}.$$

Jonathan Jedwab, Shuxing Li

Then by Examples 3.7 and 4.3,

$$\left\{S_1(y_1, y_2, y_3, y_4), S_2(y_1, y_2, y_3, y_4), S_3(y_1, y_2, y_3, y_4), S_4(y_1, y_2, y_3, y_4)\right\}$$
(4.4)

is both a (1, 4) LP-partition in $\langle y_1, y_2, y_3, y_4 \rangle - \langle y_1, y_2 \rangle \cong \mathbb{Z}_2^4 - \mathbb{Z}_2^2$ relative to $\langle y_1, y_2 \rangle \cong \mathbb{Z}_2^2$, and a (1, 4) LP-packing in $\langle y_1, y_2, y_3, y_4 \rangle = \mathbb{Z}_2^4$ relative to $\langle y_1, y_2 \rangle \cong \mathbb{Z}_2^2$.

Let $G = \mathbb{Z}_4^4 = \langle x_1, x_2, x_3, x_4 \rangle$. We firstly use Theorem 4.6 to construct a (4, 4) LP-partition in $G - \langle x_1, x_2, x_3^2, x_4^2 \rangle = \mathbb{Z}_4^4 - \mathbb{Z}_4^2 \times (2\mathbb{Z}_4)^2$ relative to $\langle x_1^2, x_2^2 \rangle \cong \mathbb{Z}_2^2$. Let $Q = \langle x_1^2, x_2^2, x_3^2, x_4^2 \rangle \cong \mathbb{Z}_2^4$, and let $G' = \langle x_1^2, x_2^2, x_3^2, x_4^2 \rangle$ so that $G/G' \cong \mathbb{Z}_2^4$. With reference to (4.3), the subgroups

$$\begin{aligned} H_0 &= \langle x_1^2, x_2^2 \rangle, \quad H_1 &= \langle x_3^2, x_4^2 \rangle, \quad H_2 &= \langle x_1^2 x_3^2, x_2^2 x_4^2 \rangle, \\ H_3 &= \langle x_2^2 x_3^2, x_1^2 x_2^2 x_4^2 \rangle, \quad H_4 &= \langle x_1^2 x_2^2 x_3^2, x_1^2 x_4^2 \rangle \end{aligned}$$

form a spread when viewed as subgroups of Q. Let

$$\begin{split} V_0 &= \langle x_1, x_2, x_3^2, x_4^2 \rangle, \quad V_1 &= \langle x_1^2, x_2^2, x_3, x_4 \rangle, \quad V_2 &= \langle x_1^2, x_2^2, x_1 x_3, x_2 x_4 \rangle, \\ V_3 &= \langle x_1^2, x_2^2, x_2 x_3, x_1 x_2 x_4 \rangle, \quad V_4 &= \langle x_1^2, x_2^2, x_1 x_2 x_3, x_1 x_4 \rangle, \end{split}$$

so that, with reference to (4.3), $\{V_0/G', V_1/G', V_2/G', V_3/G', V_4/G'\}$ is a spread of G/G' and $H_i \leq V_i$ for i = 1, 2, 3, 4. Then the factor groups $V_i/H_i, G'/H_i, Q/H_i$ take the following form for i = 1, 2, 3, 4:

i	V_i/H_i	G'/H_i	Q/H_i
1	$\langle x_1^2 H_1, x_2^2 H_1, x_3 H_1, x_4 H_1 \rangle \cong \mathbb{Z}_2^4$	$\langle x_1^2 H_1, x_2^2 H_1 \rangle \cong \mathbb{Z}_2^2$	$\langle x_1^2 H_1, x_2^2 H_1 \rangle \cong \mathbb{Z}_2^2$
2	$\langle x_1^2 H_2, x_2^2 H_2, x_1 x_3 H_2, x_2 x_4 H_2 \rangle \cong \mathbb{Z}_2^4$	$\langle x_1^2 H_2, x_2^2 H_2 \rangle \cong \mathbb{Z}_2^2$	$\langle x_1^2 H_2, x_2^2 H_2 \rangle \cong \mathbb{Z}_2^2$
3	$\langle x_1^2 H_3, x_2^2 H_3, x_2 x_3 H_3, x_1 x_2 x_4 H_3 \rangle \cong \mathbb{Z}_2^4$	$\langle x_1^2 H_3, x_2^2 H_3 \rangle \cong \mathbb{Z}_2^2$	$\langle x_1^2 H_3, x_2^2 H_3 \rangle \cong \mathbb{Z}_2^2$
4	$\langle x_1^2 H_4, x_2^2 H_4, x_1 x_2 x_3 H_4, x_1 x_4 H_4 \rangle \cong \mathbb{Z}_2^4$	$\langle x_1^2 H_4, x_2^2 H_4 \rangle \cong \mathbb{Z}_2^2$	$\langle x_1^2 H_4, x_2^2 H_4 \rangle \cong \mathbb{Z}_2^2$

Since (4.4) is a (1,4) LP-partition in $\langle y_1, y_2, y_3, y_4 \rangle - \langle y_1, y_2 \rangle$ relative to $\langle y_1, y_2 \rangle$, where $\langle y_1, y_2, y_3, y_4 \rangle = \mathbb{Z}_2^4$, it follows for i = 1, 2, 3, 4 that the 4 subsets of V_i/H_i shown in row *i* of the following table form a (1,4) LP-partition in $V_i/H_i - G'/H_i$ relative to Q/H_i :

$S_1(x_1^2, x_2^2, x_3, x_4)H_1$	$S_2(x_1^2, x_2^2, x_3, x_4)H_1$	$S_3(x_1^2, x_2^2, x_3, x_4)H_1$	$S_4(x_1^2, x_2^2, x_3, x_4)H_1$
$S_1(x_1^2, x_2^2, x_1x_3, x_2x_4)H_2$	$S_2(x_1^2, x_2^2, x_1x_3, x_2x_4)H_2$	$S_3(x_1^2, x_2^2, x_1x_3, x_2x_4)H_2$	$S_4(x_1^2, x_2^2, x_1x_3, x_2x_4)H_2$
$S_1(x_1^2, x_2^2, x_2x_3, x_1x_2x_4)H_3$	$S_2(x_1^2, x_2^2, x_2x_3, x_1x_2x_4)H_3$	$S_3(x_1^2, x_2^2, x_2x_3, x_1x_2x_4)H_3$	$S_4(x_1^2, x_2^2, x_2x_3, x_1x_2x_4)H_3$
$S_1(x_1^2, x_2^2, x_1x_2x_3, x_1x_4)H_4$	$S_2(x_1^2, x_2^2, x_1x_2x_3, x_1x_4)H_4$	$S_3(x_1^2, x_2^2, x_1x_2x_3, x_1x_4)H_4$	$S_4(x_1^2, x_2^2, x_1x_2x_3, x_1x_4)H_4$

Now regard each entry of the above table as a 12-subset of G, by interpreting H_i as the sum of four elements in G. By Theorem 4.6, summing the entries in each column of the table then gives a (4, 4) LP-partition $\{R_1, R_2, R_3, R_4\}$ in $G - V_0$ relative to H_0 .

We next use Theorem 4.4 to construct a (4,4) LP-packing in $G = \mathbb{Z}_4^4$ relative to $U = \langle x_1, x_2^2, x_3^2 \rangle \cong \mathbb{Z}_4 \times (2\mathbb{Z}_4)^2$. We have $V_0/H_0 = \langle x_1H_0, x_2H_0, x_3^2H_0, x_4^2H_0 \rangle \cong \mathbb{Z}_2^4$ and $U/H_0 = \langle x_1H_0, x_3^2H_0 \rangle \cong \mathbb{Z}_2^2$. Since (4.4) is a (1,4) LP-packing in $\langle y_1, y_2, y_3, y_4 \rangle = \mathbb{Z}_2^4$ relative to $\langle y_1, y_2 \rangle$, it follows that

$$\left\{S_1(x_1, x_3^2, x_2, x_4^2)H_0, S_2(x_1, x_3^2, x_2, x_4^2)H_0, S_3(x_1, x_3^2, x_2, x_4^2)H_0, S_4(x_1, x_3^2, x_2, x_4^2)H_0\right\}$$

COMBINATORIAL THEORY 27 (2020), #P00

is a (1,4) LP-packing in V_0/H_0 relative to U/H_0 . Regard each of these subsets as a 12-subset of G, by interpreting H_0 as the sum of four elements in G. Then by Theorem 4.4,

$$\left\{ \begin{aligned} R_1 + S_1(x_1, x_3^2, x_2, x_4^2) H_0, & R_2 + S_2(x_1, x_3^2, x_2, x_4^2) H_0, \\ R_3 + S_3(x_1, x_3^2, x_2, x_4^2) H_0, & R_4 + S_4(x_1, x_3^2, x_2, x_4^2) H_0 \end{aligned} \right\}$$

is a (4, 4) LP-packing in G relative to U.

Using the same (4, 4) LP-partition $\{R_1, R_2, R_3, R_4\}$, we can similarly use Theorem 4.4 to construct a (4, 4) LP-packing in G relative to the subgroup $U' = \langle x_1, x_2 \rangle \cong \mathbb{Z}_4^2$ and to $U'' = \langle x_1^2, x_2^2, x_3^2, x_4^2 \rangle \cong (2\mathbb{Z}_4)^4$. Following the above procedure, and using $U'/H_0 = \langle x_1H_0, x_2H_0 \rangle \cong \mathbb{Z}_2^2$ and $U''/H_0 = \langle x_3^2H_0, x_4^2H_0 \rangle \cong \mathbb{Z}_2^2$, shows that

$$\left\{ \begin{aligned} R_1 + S_1(x_1, x_2, x_3^2, x_4^2) H_0, & R_2 + S_2(x_1, x_2, x_3^2, x_4^2) H_0, \\ R_3 + S_3(x_1, x_2, x_3^2, x_4^2) H_0, & R_4 + S_4(x_1, x_2, x_3^2, x_4^2) H_0 \end{aligned} \right\}$$

is a (4, 4) LP-packing in G relative to U', and that

$$\left\{ \begin{aligned} R_1 + S_1(x_3^2, x_4^2, x_1, x_2) H_0, & R_2 + S_2(x_3^2, x_4^2, x_1, x_2) H_0, \\ R_3 + S_3(x_3^2, x_4^2, x_1, x_2) H_0, & R_4 + S_4(x_3^2, x_4^2, x_1, x_2) H_0 \end{aligned} \right\}$$

is a (4, 4) LP-packing in G relative to U''.

5. Recursive Construction of LP-partitions and LP-packings

In this section, we recursively construct infinite families of LP-partitions and LP-packings, as shown schematically in Figure 5.1. On the right side of the figure, we iteratively use Theorem 4.6 to produce an LP-partition in the group $\mathbb{Z}_{p^a}^{2s} - \mathbb{Z}_{p^a}^s \times (p\mathbb{Z}_{p^a})^s$ relative to $H_0 \cong \mathbb{Z}_p^s$ by lifting with respect to $H_i \cong \mathbb{Z}_p^s$ an LP-partition in a factor group, for each $i = 1, \ldots, p^s$ (see Theorem 5.2). On the left side of the figure, we iteratively use Theorem 4.4 to produce an LP-packing in $\mathbb{Z}_{p^a}^{2s}$ relative to an order p^{as} subgroup, by lifting with respect to H_0 an LP-packing in a factor group and taking the union of the resulting subsets with the subsets of the LP-partition at the same height (see Theorem 5.3).

We begin with a technical lemma.

Lemma 5.1. Let p be prime, and let s and a_1, \ldots, a_{2s} be positive integers. Let $G = \prod_{j=1}^{2s} \mathbb{Z}_{p^{a_j}}$, and let Q be the unique subgroup of G isomorphic to \mathbb{Z}_p^{2s} . Let H_0, \ldots, H_{p^s} be a spread of Q, where (without loss of generality) H_0 is contained in the first s direct factors of G. Then for each $i = 1, \ldots, p^s$, we have $G/H_i \cong \prod_{j=1}^s \mathbb{Z}_{p^{a_j}} \times \prod_{j=s+1}^{2s} \mathbb{Z}_{p^{a_j-1}}$, and the quotient group Q/H_i is contained in the first s direct factors of G/H_i .

Proof. Fix $i \in \{1, \ldots, p^s\}$. Since $Q = H_0H_i$, we can choose a set of generators for Q (depending on i) so that H_i is contained in the last s direct factors of Q. Therefore $G/H_i \cong \prod_{j=1}^s \mathbb{Z}_{p^{a_j}} \times \prod_{j=s+1}^{2s} \mathbb{Z}_{p^{a_j-1}}$. Let the first s direct factors of G be $\langle x_1, \ldots, x_s \rangle$, so that the first s direct factors of G/H_i are $\langle x_1H_i, \ldots, x_sH_i \rangle$. Since $H_0 = \langle x_1^{p^{a_1-1}}, \ldots, x_s^{p^{a_s-1}} \rangle$, we have $Q/H_i = \langle x_1^{p^{a_1-1}}H_i, \ldots, x_s^{p^{a_s-1}}H_i \rangle$. Therefore Q/H_i is contained in the first s direct factors of G/H_i .



Figure 5.1: Recursive construction of LP-packings and LP-partitions according to Theorems 5.2 and 5.3

We now apply Theorem 4.6 iteratively to produce an infinite family of LP-partitions.

Theorem 5.2. Let p be prime, let a, s be positive integers, and let $G = \mathbb{Z}_{p^a}^{2s}$. Let H_0 and V_0 be the unique subgroups of G for which $H_0 \cong \mathbb{Z}_p^s$ is contained in the first s direct factors of G and $V_0 \cong \mathbb{Z}_{p^a}^s \times \mathbb{Z}_{p^{a-1}}^s$ satisfies $V_0/H_0 \cong \mathbb{Z}_{p^{a-1}}^{2s}$. Then there exists a $(p^{(a-1)s}, p^s)$ LP-partition in $G - V_0$ relative to H_0 .

Proof. The proof is by induction on $a \ge 1$. The case a = 1 is given by Example 4.3. Suppose that all cases up to $a - 1 \ge 1$ are true. Let Q and G' be the unique subgroups of G for which $Q \cong \mathbb{Z}_p^{2s}$ and $G' \cong \mathbb{Z}_{p^{a-1}}^{2s}$. Since $G/G' \cong \mathbb{Z}_p^{2s}$, by Example 3.7 there exists a spread in G/G' one of whose elements we may take to be V_0/G' . Let the remaining elements of this spread be $V_1/G', \ldots, V_{p^s}/G'$, where V_1, \ldots, V_{p^s} are subgroups of G containing G'. Let ϕ be a group isomorphism from G/G' to Q for which $\phi(V_0/G') = H_0$, and let $H_i = \phi(V_i/G')$ for i = 0.

COMBINATORIAL THEORY 27 (2020), #P00

1,..., p^s . Since $\{V_0/G', \ldots, V_{p^s}/G'\}$ is a spread of G/G', it follows that $\{H_0, \ldots, H_{p^s}\}$ is a spread of Q. Furthermore, for each i we have $H_i \leq Q \leq G' \leq V_i$ and therefore H_i is a subgroup of V_i , and $V_i/H_i \cong G'$.

Let $i \in \{1, ..., p^s\}$. By construction, we have $H_i \leq V_i$ and $V_i/H_i \cong G' \cong \mathbb{Z}_{p^{a-1}}^{2s}$. By Lemma 5.1, we have $G'/H_i \cong \mathbb{Z}_{p^{a-1}}^s \times \mathbb{Z}_{p^{a-2}}^s$ and $Q/H_i \cong \mathbb{Z}_p^s$, and Q/H_i is contained in the first *s* direct factors of V_i/H_i . We also have $(G'/H_i)/(Q/H_i) \cong G'/Q \cong \mathbb{Z}_{p^{a-2}}^{2s}$. Therefore by the inductive hypothesis there exists a $(p^{(a-2)s}, p^s)$ LP-partition in $V_i/H_i - G'/H_i$ relative to Q/H_i . Then by Theorem 4.6 there exists a $(p^s \cdot p^{(a-2)s}, p^s)$ LP-partition in $G - V_0$ relative to H_0 , so the case *a* is true.

We now apply Theorem 4.4 iteratively, making use of the LP-partitions constructed in Theorem 5.2, to produce an infinite family of LP-packings in groups of increasing exponent relative to an arbitrary subgroup of the appropriate order. The initial LP-packing in an elementary abelian group is obtained from a spread.

Theorem 5.3. Let p be prime, and let a, s be positive integers. Let $G = \mathbb{Z}_{p^a}^{2s}$ and let U be a subgroup of G of order p^{as} . Then there exists a $(p^{(a-1)s}, p^s)$ LP-packing in G relative to U.

Proof. The proof is by induction on $a \ge 1$. The case a = 1 follows from the construction of a spread in Example 3.7, because by a suitable choice of generators of $G = \mathbb{Z}_p^{2s}$ we may assume that the element H_0 of the spread is U.

Suppose that all cases up to $a - 1 \ge 1$ are true. Since U is a subgroup of G of order p^{as} , it has at most s direct factors isomorphic to \mathbb{Z}_{p^a} and has rank at least s. We may therefore choose $V_0 \cong \mathbb{Z}_{p^a}^s \times \mathbb{Z}_{p^{a-1}}^s$ for which $U \le V_0 \le G$, and also assume that the subgroup of V_0 isomorphic to $\mathbb{Z}_{p^a}^s$ is contained in the first s direct factors of G and has intersection of rank s with U. Therefore U contains a subgroup $H_0 \cong \mathbb{Z}_p^s$ for which H_0 is contained in the first s direct factors of G. We then have $V_0/H_0 \cong \mathbb{Z}_{p^{a-1}}^{2s}$ and $|U/H_0| = p^{(a-1)s}$.

Then by Theorem 5.2, there exists a $(p^{(a-1)s}, p^s)$ LP-partition in $G - V_0$ relative to H_0 . By the inductive hypothesis, there exists a $(p^{(a-2)s}, p^s)$ LP-packing in V_0/H_0 relative to U/H_0 . Therefore by Theorem 4.4, there exists a $(p^{(a-1)s}, p^s)$ LP-packing in G relative to U, so the case a is true.

The construction process illustrated in Figure 5.1 shares many features with the recursive construction of difference sets from relative difference sets introduced in [33], which is illustrated in [34] using an analogous representation to Figure 5.1. In both settings, an ingredient on the left side is lifted from a factor group and combined with an ingredient on the right side to form a larger example on the left side; and multiple instances of an ingredient on the right side. The ingredients in [33] corresponding to an LP-packing and an LP-partition are, respectively, a covering extended building set for constructing a difference set, and a building set for constructing a relative difference set. However, in [33] the ingredients are placed into distinct cosets of a subgroup and so can be combined without regard to their intersections as subsets of the subgroup. In contrast, in the construction of Theorem 4.4, the lifted subset P'_j of an LP-packing is combined with the subset R_j of an LP-partition by taking their union. Likewise, in the construction

of Theorem 4.6, the subsets of the i^{th} LP-partition are lifted to S'_{i1}, \ldots, S'_{it} , and then subset R_j of the new LP-partition is formed as the union of S'_{1j}, \ldots, S'_{tj} (with one lifted subset derived from each of the *t* LP-partitions). This leads to an additional constraint, that the subsets P'_j and S'_{1j}, \ldots, S'_{tj} must be disjoint for each *j*.

We now extend Theorem 5.3 using constructions from Section 3 in order to obtain the result stated in Theorem 1.1.

Corollary 5.4. Let p be prime, let s_1, \ldots, s_v be nonnegative integers (not all zero), and let $m = \min\{s_i \mid s_i > 0\}$. For each $i = 1, \ldots, v$, let $G_i = \mathbb{Z}_{p^i}^{2s_i}$ and let U_i be a subgroup of G_i of order p^{is_i} . Let $G = \prod_{i=1}^v G_i$, let $U = \prod_{i=1}^v U_i$, and let $n = \sqrt{|G|}$. Then for each $j = 0, \ldots, m-1$, there exists an $\left(\frac{n}{p^{m-j}}, p^{m-j}\right)$ LP-packing in G relative to U.

Proof. For each i = 1, ..., v for which $s_i > 0$, by Theorem 5.3 there exists a $(p^{(i-1)s_i}, p^{s_i})$ LP-packing in G_i relative to U_i . By a straightforward induction, Corollary 3.17 then gives an $\left(\frac{n}{p^m}, p^m\right)$ LP-packing in G relative to U. For $j \in \{0, ..., m-1\}$, apply Lemma 3.11 with $s = p^j$ to give an $\left(\frac{n}{p^{m-j}}, p^{m-j}\right)$ LP-packing in G relative to U.

Remark 5.5. By Definition 3.1, the constructed LP-packing in Corollary 5.4 is a collection of p^{m-j} disjoint regular $\left(n, \frac{n}{p^{m-j}}\right)$ Latin square type PDSs in G avoiding U. By Lemma 3.9, we also obtain a regular Latin square type PDS in G with parameters (n, r) and (n, r+1), for each positive integer multiple r of $\frac{n}{p^{m-j}}$ satisfying $r \leq n$.

This improves on numerous previous constructions of PDSs in three respects: by constructing a collection of disjoint PDSs in G where previously only a single PDS was constructed; by identifying the order n subgroup U and showing it can be chosen arbitrarily; and by producing Latin square type PDSs with new parameters (n, r). We give an illustrative selection of examples.

- (i) Result 2.7 constructs a single regular (n, r) Latin square type PDS in G for each positive integer $r \leq p^m + 1$, whereas the case j = 0 of Corollary 5.4 gives a disjoint collection of p^m such PDSs with $r = \frac{n}{n^m}$.
- (*ii*) Result 2.8 (*iii*) constructs a regular (n, r) Latin square type PDS in G for each positive integer multiple r of $\frac{n}{p^{\operatorname{gcd}(s_1,\ldots,s_v)}}$ satisfying $r \leq n$, whereas the case j = 0 of Corollary 5.4 gives such a PDS for each positive integer multiple r of $\frac{n}{p^m}$ satisfying $r \leq n$ and whose parameters therefore differ when $m > \operatorname{gcd}(s_1,\ldots,s_v)$.
- (*iii*) The LP-packing of [82, Cor. 6.2] described in Table 1.1 is a disjoint collection of p regular $(n, \frac{n}{p})$ PDSs in certain groups G constrained to have at most one direct factor with an exponent that is an odd power of p, whereas the case j = m 1 of Corollary 5.4 gives such disjoint collections of PDSs without this constraint.
- (*iv*) In each group for which the lifting approach in [48], [50, Sects. 4, 5], [52] constructs a single PDS, Corollary 5.4 provides a collection of disjoint PDSs having the same parameters and avoiding an arbitrary order n subgroup U of G.

(v) Result 2.9 constructs a partition of $G - 1_G$ into p regular PDSs in G, of which p - 1 are of $(n, \frac{n}{p})$ Latin square type and one is of $(n, \frac{n}{p} + 1)$ Latin square type, where G is constrained to have $s_{2i+1} = 0$ for each i > 0. In contrast, by applying Remark 3.10 (i) to the case j = 0 of Corollary 5.4 we obtain a partition of $G - 1_G$ into p^m regular PDSs in G, of which $p^m - 1$ are of $(n, \frac{n}{p^m})$ Latin square type and one is of $(n, \frac{n}{p^m} + 1)$ Latin square type, without constraining G.

The LP-packings in Corollary 5.4 are all contained in *p*-groups of the form $H \times H$, having even rank. We now extend this result to *p*-groups of odd rank, starting from the following infinite family of odd rank examples that was obtained from an elegant construction using Galois rings. (We could likewise make use of the further odd rank example of Example 3.14 (*ii*), relative to a nonelementary abelian group.)

Result 5.6 ([51, Lem. 3.1]). Let p be prime and let w > 1 be an odd integer. Then there exists a $(p^{w-\sigma(p,w)}, p^{\sigma(p,w)})$ LP-packing in $\mathbb{Z}_{p^2}^w$ relative to $(p\mathbb{Z}_{p^2})^w$, where the function $\sigma(p, w)$ is defined in (2.2).

Corollary 5.7. Let p be prime, let w > 1 be an odd integer, and let s_1, \ldots, s_v be nonnegative integers (possibly all zero). Let $m = \min(\{\sigma(p, w)\} \cup \{s_i \mid s_i > 0\})$. For each $i = 1, \ldots, v$, let $G_i = \mathbb{Z}_{p^i}^{2s_i}$ and let U_i be a subgroup of G_i of order p^{is_i} . Let $G = \mathbb{Z}_{p^2}^w \times \prod_{i=1}^v G_i$, let $U = (p\mathbb{Z}_{p^2})^w \times \prod_{i=1}^v U_i$, and let $n = \sqrt{|G|}$. Then for each $j = 0, \ldots, m-1$, there exists an $\left(\frac{n}{p^{m-j}}, p^{m-j}\right)$ LP-packing in G relative to U.

Proof. In the case that the s_i are all zero, we have $m = \sigma(p, w)$ and the result follows by applying Lemma 3.11 with $s = p^j$ to Result 5.6. Otherwise, at least one of the s_i is positive and we may define $m' = \min\{s_i \mid s_i > 0\}$. The case j = 0 of Corollary 5.4 then gives an $\left(\frac{n/p^w}{p^{m'}}, p^{m'}\right)$ LP-packing in $\prod_{i=1}^{v} G_i$ relative to $\prod_{i=1}^{v} U_i$. Combine with the LP-packing of Result 5.6 using Corollary 3.17 to give an $\left(\frac{n}{p^m}, p^m\right)$ LP-packing in G relative to U. Then for $j \in \{0, \ldots, m-1\}$, apply Lemma 3.11 with $s = p^j$ to give an $\left(\frac{n}{p^{m-j}}, p^{m-j}\right)$ LP-packing in G relative to U.

Remark 5.8. By Definition 3.1, the constructed LP-packing in Corollary 5.7 is a collection of p^{m-j} disjoint regular $\left(n, \frac{n}{p^{m-j}}\right)$ Latin square type PDSs in *G* avoiding *U*. By Lemma 3.9, we also obtain a regular Latin square type PDS in *G* with parameters (n, r) and with parameters (n, r+1), for each positive integer multiple *r* of $\frac{n}{p^{m-j}}$ satisfying $r \leq n$.

We can show that this improves on previous constructions along similar lines to Remark 5.5. For example, Result 2.8 (*ii*) constructs a single regular (n, r) Latin square type PDS in G for v = 1 and i = 1 and s_1 a positive integer multiple of w, for each positive integer multiple r of $\frac{n}{p^{\sigma(p,w)}}$ satisfying $r \leq n$. In contrast, the case j = 0 of Corollary 5.7 gives a disjoint collection of p^m such PDSs in much more general groups G (having arbitrary v and s_i).

6. NLP-packings

In this section, we introduce a (c, t - 1) NLP-packing as a collection of t - 1 disjoint regular (tc, c) negative Latin square type PDSs, in an abelian group of order t^2c^2 , whose union satisfies

an additional property. This structure forms a counterpart to a (c, t) LP-packing. We provide examples of NLP-packings, and show how to combine them with LP-packings via a product construction to form NLP-packings in the direct product of the starting groups. In conjunction with the LP-packings of Corollaries 5.4 and 5.7, this produces infinite families of NLP-packings. However, we do not have a lifting result for NLP-packings similar to Theorem 4.4.

Definition 6.1 (NLP-packing). Let t > 1 and c > 0 be integers, and let G be an abelian group of order t^2c^2 . A (c, t - 1) *NLP-packing in* G is a collection $\{P_1, \ldots, P_{t-1}\}$ of t - 1 regular (tc, c) negative Latin square type PDSs in G for which $G - 1_G - \sum_{i=1}^{t-1} P_i$ is a regular (tc, c-1) negative Latin square type PDS in G.

Remark 6.2.

- (i) We can rephrase Definition 6.1 to say that a (c, t 1) NLP-packing in G is a partition of $G - 1_G$ into t regular partial difference sets, of which t - 1 are of (tc, c) negative Latin square type and one is of (tc, c - 1) negative Latin square type. Previous results on NLP-packings use this phrasing.
- (*ii*) In Definition 6.1, each P_i is a c(tc+1)-subset of G, and $G-1_G-\sum_{i=1}^{t-1}P_i$ is a (c-1)(tc+1)-subset of G, so the subsets P_i of a (c, t-1) NLP-packing are necessarily disjoint. For all $c \ge 1$, if a (c, t-1) NLP-packing exists in G, then it contains the maximum possible number of disjoint regular (tc, c) negative Latin square type PDSs.
- (*iii*) The case r = 0 was allowed in Definition 2.4 so that the definition of a (c, t 1) NLP-packing applies when c = 1.

NLP-packings have been studied in a series of previous papers, as summarized in Table 1.2. We now characterize a (c, t - 1) NLP-packing using character sums.

Lemma 6.3. Let P_1, \ldots, P_{t-1} be disjoint c(tc+1)-subsets of an abelian group G of order t^2c^2 . Then $\{P_1, \ldots, P_{t-1}\}$ is a (c, t-1) NLP-packing in G if and only if the multiset equality

$$\{\chi(P_1), \dots, \chi(P_{t-1})\} = \{c, \dots, c\} \text{ or } \{-(t-1)c, c, \dots, c\}$$

holds for all nonprincipal characters χ of G.

Proof. Each P_i is a c(tc+1)-subset of G, and the P_i are disjoint, so $G - 1_G - \sum_{i=1}^{t-1} P_i$ is a (c-1)(tc+1)-subset of G. Lemma 2.5 then shows that $\{P_1, \ldots, P_{t-1}\}$ is a (c, t-1) NLP-packing in G if and only if

$$\chi(P_i) \in \{c, -(t-1)c\} \text{ and } \chi\left(G - 1_G - \sum_{i=1}^{t-1} P_i\right) \in \{c-1, -(tc-c+1)\}$$
(6.1)

for all nonprincipal characters χ of G. For each nonprincipal character χ of G, condition (6.1) is equivalent to

$$\chi(P_i) \in \{c, -(t-1)c\}$$
 and $\sum_{i=1}^{t-1} \chi(P_i) \in \{-c, (t-1)c)\},\$

which in turn is equivalent to

$$\{\chi(P_1),\ldots,\chi(P_{t-1})\} = \{c,\ldots,c\} \text{ or } \{-(t-1)c,c,\ldots,c\},\$$

as required.

Lemmas 2.5 and 6.3 allow us to combine subsets from an NLP-packing to obtain negative Latin square type PDSs with various parameters.

Lemma 6.4. Suppose that $\{P_1, \ldots, P_{t-1}\}$ is a (c, t-1) NLP-packing in an abelian group G of order t^2c^2 , and let I be a b-subset of $\{1, \ldots, t-1\}$. Then $\sum_{i \in I} P_i$ is a regular (tc, bc) negative Latin square type PDS in G.

Remark 6.5. Suppose that $\{P_1, \ldots, P_{t-1}\}$ is a (c, t-1) NLP-packing in an abelian group G of order t^2c^2 .

- (*i*) Let *I* be a *b*-subset of $\{1, \ldots, t-1\}$. By Lemma 2.5, an r(n+1)-subset *D* of an abelian group *G* of order n^2 is a regular (n, r) negative Latin square type PDS in *G* if and only if $G 1_G D$ is a regular (n, n r 1) negative Latin square type PDS in *G*. Lemma 6.4 then shows that $G 1_G \sum_{i \in I} P_i$ is a regular (tc, (t-b)c 1) negative Latin square type PDS in *G*.
- (*ii*) For each *i*, let $\widetilde{P_i} = \{\chi \in \widehat{G} \mid \chi(P_i) = -(t-1)c\}$. Let $P = G 1_G \sum_{i=1}^{t-1} P_i$ and let $\widetilde{P} = \{\chi \in \widehat{G} \mid \chi(P) = -(tc c + 1)\}$. By Lemma 2.5 and Definition 6.1, each $\widetilde{P_i}$ is a regular (tc, c) negative Latin square type PDS in \widehat{G} , and \widetilde{P} is a regular (tc, c-1) negative Latin square type PDS in \widehat{G} . By Lemma 6.3, each nonprincipal character χ of G belongs to exactly one of $\widetilde{P_1}, \ldots, \widetilde{P_{t-1}}, \widetilde{P}$. By Definition 6.1, the collection $\{\widetilde{P_1}, \ldots, \widetilde{P_{t-1}}\}$ is therefore a (c, t-1) NLP-packing in \widehat{G} .

We give some small examples of NLP-packings.

Example 6.6.

(i) [82, p. 373]. A (1,3) NLP-packing $\{P_1, P_2, P_3\}$ in $\mathbb{Z}_2^4 = \langle x, y, z, w \rangle$ is given by

$$P_1 = x + z + yw + xw + yz,$$

$$P_2 = y + w + xyzw + yzw + xyw,$$

$$P_3 = xy + zw + xz + xyz + xzw.$$

(*ii*) [82, p. 373]. A (1,3) NLP-packing $\{P_1, P_2, P_3\}$ in $\mathbb{Z}_4^2 = \langle x, y \rangle$ is given by

$$P_{1} = x^{2} + y + y^{3} + xy + x^{3}y^{3},$$

$$P_{2} = x + x^{3} + y^{2} + xy^{3} + x^{3}y,$$

$$P_{3} = xy^{2} + x^{3}y^{2} + x^{2}y + x^{2}y^{3} + x^{2}y^{2}.$$

(*iii*) [82, Ex. 1]. A (1, 2) NLP-packing $\{P_1, P_2\}$ in $\mathbb{Z}_3^2 = \langle x, y \rangle$ is given by

$$P_{1} = x + x^{2} + y + y^{2},$$

$$P_{2} = xy + x^{2}y^{2} + xy^{2} + x^{2}y.$$

We next show that combining the subsets of an NLP-packing into equally-sized collections gives an NLP-packing with fewer subsets.

Lemma 6.7. Suppose there exists a (c, t - 1) NLP-packing in an abelian group G of order t^2c^2 , and suppose s divides t. Then there exists an $(sc, \frac{t}{s} - 1)$ NLP-packing in G.

Proof. Let $\{P_1, \ldots, P_{t-1}\}$ be a (c, t-1) NLP-packing in G, and let

$$P'_i = P_{is+1} + P_{is+2} + \dots + P_{is+s}$$
 for $i = 0, 1, \dots, \frac{t}{s} - 2$.

Then Lemma 6.3 shows that $\{P'_0, \ldots, P'_{\frac{t}{s}-2}\}$ is an $(sc, \frac{t}{s}-1)$ NLP-packing in G.

The following product construction combines an NLP-packing with an LP-packing to form an NLP-packing in a larger group.

Theorem 6.8 ([82, Thm. 2.1]). Suppose there exists a $(c_1, t - 1)$ NLP-packing in an abelian group G_1 of order $t^2c_1^2$, and a (c_2, t) LP-packing in an abelian group G_2 of order $t^2c_2^2$ relative to a subgroup U_2 of order tc_2 . Then there exists a $(tc_1c_2, t - 1)$ NLP-packing in $G_1 \times G_2$.

Proof. Let $\{P_{1,0}, \ldots, P_{1,t-2}\}$ be a $(c_1, t-1)$ NLP-packing in G_1 , and let $\{P_{2,0}, \ldots, P_{2,t-1}\}$ be a (c_2, t) LP-packing in G_2 relative to U_2 . Define $P_{1,t-1} = G_1 - \sum_{i=0}^{t-2} P_{1,i}$ and

$$K_{\ell} = P_{1,\ell}U_2 + \sum_{i=0}^{t-1} P_{1,i}P_{2,i+\ell}$$
 for $\ell = 0, \dots, t-2$,

where the subscript $i+\ell$ is reduced modulo t. We shall use Lemma 6.3 to show that $\{K_0, K_1, \ldots, K_{t-2}\}$ is a $(tc_1c_2, t-1)$ NLP-packing in $G_1 \times G_2$.

The $P_{1,i}$ are disjoint $c_1(tc_1 + 1)$ -subsets of G_1 for $i = 0, \ldots, t-2$, and by definition $P_{1,t-1}$ is a $c_1(tc_1 - t + 1)$ -subset of G_1 disjoint from the other $P_{1,i}$. The $P_{2,i}$ are $c_2(tc_2 - 1)$ -subsets of $G_2 - U_2$. Therefore the K_ℓ are disjoint subsets of $G_1 \times G_2$, each of size $c_1(tc_1 + 1)tc_2 + c_1(tc_1 - t + 1)c_2(tc_2 - 1) + (t - 1)c_1(tc_1 + 1)c_2(tc_2 - 1) = tc_1c_2(t^2c_1c_2 + 1)$. Let χ be a nonprincipal character of $G_1 \times G_2$, and let $\chi_j = \chi|_{G_i}$ for j = 1, 2. Then

$$\chi(K_{\ell}) = \chi_1(P_{1,\ell})\chi_2(U_2) + \sum_{i=0}^{t-1} \chi_1(P_{1,i})\chi_2(P_{2,i+\ell}) \text{ for } \ell = 0, \dots, t-2.$$

By Lemma 6.3, it remains to prove that

$$\{\chi(K_0),\ldots,\chi(K_{t-2})\} = \{tc_1c_2,\ldots,tc_1c_2\} \text{ or } \{-t(t-1)c_1c_2,tc_1c_2,\ldots,tc_1c_2\}.$$
 (6.2)

Combinatorial theory **27** (2020), #P00

We are given that

$$\{\chi_1(P_{1,0}), \dots, \chi_1(P_{1,t-2})\} = \begin{cases} \{c_1(tc_1+1), \dots, c_1(tc_1+1)\} & \text{if } \chi_1 \in G_1^{\perp}, \\ \{c_1, \dots, c_1\} \text{ or } \{-(t-1)c_1, c_1, \dots, c_1\} & \text{if } \chi_1 \notin G_1^{\perp} \end{cases}$$
(6.3)

by Lemma 6.3, and that

$$\{\chi_2(P_{2,0}), \dots, \chi_2(P_{2,t-1})\} = \begin{cases} \{c_2(tc_2-1), \dots, c_2(tc_2-1)\} & \text{if } \chi_2 \in G_2^{\perp}, \\ \{-c_2, \dots, -c_2\} & \text{if } \chi_2 \in U_2^{\perp} \setminus G_2^{\perp}, \\ \{(t-1)c_2, -c_2, \dots, -c_2\} & \text{if } \chi_2 \notin U_2^{\perp} \end{cases}$$

by Lemma 3.3. The following two cases together establish (6.2).

Case 1: χ_2 is principal on U_2 .

Then $\chi_2(P_{2,i})$ is constant over *i* (regardless of whether χ_2 is principal or nonprincipal on G_2), so $\chi(K_\ell) = \chi_1(P_{1,\ell})tc_2 + \chi_1(\sum_{i=0}^{t-1} P_{1,i})\chi_2(P_{2,1}) = \chi_1(P_{1,\ell})tc_2 + \chi_1(G_1)\chi_2(P_{2,1})$. In the case that χ_1 is principal on G_1 , and therefore χ_2 is nonprincipal on G_2 , we obtain $\chi(K_\ell) = c_1(tc_1 + 1)tc_2 + t^2c_1^2(-c_2) = tc_1c_2$ for each ℓ . Otherwise χ_1 is nonprincipal on G_1 , and then we obtain $\chi(K_\ell) = \chi_1(P_{1,\ell})tc_2$ so that $\{\chi(K_0), \ldots, \chi(K_{t-2})\} = \{tc_1c_2, \ldots, tc_1c_2\}$ or $\{-(t-1)tc_1c_2, tc_1c_2, \ldots, tc_1c_2\}$.

Case 2: χ_2 is nonprincipal on U_2 .

Then $\chi(K_\ell) = \sum_{i=0}^{t-1} \chi_1(\tilde{P}_{1,i}) \chi_2(P_{2,i+\ell})$. By definition of $P_{1,t-1}$, we can extend (6.3) to obtain

$$\{\chi_1(P_{1,0}), \dots, \chi_1(P_{1,t-1})\}$$

$$= \begin{cases} \{c_1(tc_1 - t + 1), c_1(tc_1 + 1), \dots, c_1(tc_1 + 1)\} & \text{if } \chi_1 \in G_1^{\perp}, \\ \{-(t-1)c_1, c_1, \dots, c_1\} & \text{if } \chi_1 \notin G_1^{\perp}, \end{cases}$$

which can be written as

$$\{\chi_1(P_{1,0}), \dots, \chi_1(P_{1,t-1})\} = \{d - tc_1, d, \dots, d\}$$
 for all χ_1

(where the integer d equals $c_1(tc_1 + 1)$ if χ_1 is principal on G_1^{\perp} , and equals c_1 otherwise). Therefore the multiset $\{\chi(K_0), \ldots, \chi(K_{t-2})\}$ contains either one or zero occurrences of the value $(d - tc_1)(t - 1)c_2 + (t - 1)d(-c_2) = -t(t - 1)c_1c_2$, and (respectively) t - 2 or t - 1 occurrences of the value $(d - tc_1)(-c_2) + d(t - 1)c_2 + (t - 2)d(-c_2) = tc_1c_2$.

Remark 6.9. The construction of Theorem 6.8 is a restatement (without restricting t to be a prime power) of the product construction for collections of negative Latin square type PDSs and Latin square type PDSs given in [82, Thm. 2.1]. Indeed, the configuration described in [82, Lemma 2.1] for e = -1 can be represented as $\{P_1, \ldots, P_{t-1}, G - 1_G - \sum_{i=1}^{t-1} P_i\}$, where $\{P_1, \ldots, P_{t-1}\}$ is a (c, t - 1) NLP-packing in G. Although our formulation explicitly identifies the subgroup U_2 of the starting LP-packing, this does not seem to confer an advantage because the subgroup does not appear in the conclusion of the theorem.

We now extend the construction of Theorem 6.8 using Lemmas 3.11 and 6.7.

Corollary 6.10. Suppose there exists a $(c_1, t_1 - 1)$ NLP-packing in an abelian group G_1 of order $t_1^2c_1^2$, and there exists a (c_2, t_2) LP-packing in an abelian group G_2 of order $t_2^2c_2^2$ relative to a subgroup U_2 of order t_2c_2 .

- (i) Suppose that t_1 divides t_2 . Then there exists a $(t_2c_1c_2, t_1 1)$ NLP-packing in $G_1 \times G_2$.
- (*ii*) Suppose that t_2 divides t_1 . Then there exists a $(t_1c_1c_2, t_2 1)$ NLP-packing in $G_1 \times G_2$.

Proof.

- (i) By Lemma 3.11 with $s = \frac{t_2}{t_1}$, there exists a $(\frac{t_2}{t_1}c_2, t_1)$ LP-packing in G_2 relative to U_2 . Then by Theorem 6.8, there exists a $(t_2c_1c_2, t_1 - 1)$ NLP-packing in $G_1 \times G_2$.
- (*ii*) By Lemma 6.7 with $s = \frac{t_1}{t_2}$, there exists a $(\frac{t_1}{t_2}c_1, t_2 1)$ NLP-packing in G_1 . Then by Theorem 6.8, there exists a $(t_1c_1c_2, t_2 1)$ NLP-packing in $G_1 \times G_2$.

The product constructions of Theorems 3.15 and 6.8 both combine two packings to form a packing in the direct product of the starting groups. Theorem 3.15 combines two LP-packings to form an LP-packing, whereas Theorem 6.8 combines an NLP-packing and an LP-packing to form an NLP-packing. We note in passing that we can likewise combine two NLP-packings to form a collection of regular Latin square type PDSs that partitions the nonidentity elements of the product group.

Proposition 6.11 ([82, Thm. 2.3]). For j = 1, 2, suppose there exists a $(c_j, t-1)$ NLP-packing in an abelian group G_j of order $t^2c_j^2$. Then there exists a collection $\{K_1, \ldots, K_{t-1}\}$ of t-1 disjoint regular $(t^2c_1c_2, tc_1c_2)$ Latin square type PDSs in $G_1 \times G_2$ for which $G_1 \times G_2 - 1_{G_1 \times G_2} - \sum_{\ell=1}^{t-1} K_\ell$ is a regular $(t^2c_1c_2, tc_1c_2 + 1)$ Latin square type PDS in $G_1 \times G_2$.

Proof (Outline). For j = 1, 2, let $\{P_{j,0}, \ldots, P_{j,t-2}\}$ be a $(c_j, t-1)$ NLP-packing in G_j and define $P_{j,t-1} = G_j - \sum_{i=0}^{t-2} P_{j,i}$. Define

$$K_{\ell} = \sum_{i=0}^{t-1} P_{1,i} P_{2,i+\ell}$$
 for $\ell = 1, \dots, t-1$,

where the subscript $i+\ell$ is reduced modulo t. Model the rest of the proof on that of Theorem 6.8, distinguishing the case that $\chi|_{G_1}$ is principal on G_1 from the case that $\chi|_{G_1}$ is nonprincipal on G_1 , and applying Lemma 2.5.

Remark 6.12. In the construction of Proposition 6.11, if we could identify an order $t^2c_1c_2$ subgroup U of $G_1 \times G_2$ contained in $G_1 \times G_2 - \sum_{\ell=1}^{t-1} K_\ell$ for which $G_1 \times G_2 - U - \sum_{\ell=1}^{t-1} K_\ell$ is a regular $(t^2c_1c_2, tc_1c_2)$ Latin square type PDS in $G_1 \times G_2$, then we could conclude that $\{K_1, \ldots, K_{t-1}, G_1 \times G_2 - U - \sum_{\ell=1}^{t-1} K_\ell\}$ is a (tc_1c_2, t) LP-packing in $G_1 \times G_2$ relative to U. However, we do not know how to identify such a subgroup U (see also Remark 7.3). We now apply Theorem 6.8 to the small NLP-packings of Example 6.6 and LP-packings from Corollaries 5.4 and 5.7 to produce two infinite families of NLP-packings.

Corollary 6.13. Let u, w be nonegative integers. Let s_3, \ldots, s_v be nonnegative integers (possibly all zero).

- (i) Let $u \neq 1$, let $w \notin \{1, 2, 3\}$ and let each $s_i \neq 1$. Let G be either $\mathbb{Z}_2^{2u+4} \times \mathbb{Z}_4^w \times \prod_{i=3}^v \mathbb{Z}_{2^i}^{2s_i}$ or $\mathbb{Z}_2^{2u} \times \mathbb{Z}_4^{w+2} \times \prod_{i=3}^v \mathbb{Z}_{2^i}^{2s_i}$, and let $n = \sqrt{|G|}$. Then there exists an $(\frac{n}{4}, 3)$ NLP-packing in G.
- (ii) Let $u \neq 0$ and let $w \neq 1$. Let $G = \mathbb{Z}_3^{2u} \times \mathbb{Z}_9^w \times \prod_{i=3}^v \mathbb{Z}_{3^i}^{2s_i}$ and let $n = \sqrt{|G|}$. Then there exists an $\left(\frac{n}{3}, 2\right)$ NLP-packing in G.

Proof.

(i) In the case that u, w, and each s_i is zero, the result is given by Example 6.6 (i), (ii). Otherwise, let G' = Z₂^{2u} × Z₄^w × Π^v_{i=3} Z_{2ⁱ}<sup>2s_i</sub> and construct an (^{n/4}/₄, 4) LP-packing in G' relative to a subgroup of order n/4. To do so when w is even, take p = 2 and (s₁, s₂) = (u, ^w/₂) and j = m - 2 ≥ 0 in Corollary 5.4. To do so when w is odd, note that σ(2, w) = ^{w-1}/₂ ≥ 2 and take p = 2 and (s₁, s₂) = (u, 0) and j = m - 2 ≥ 0 in Corollary 5.7. Now let H be either Z₂⁴ or Z₄². Use Theorem 6.8 to combine the (1, 3) NLP-packing in H given by Example 6.6 (i) (ii) with the constructed (^{n/4}/₄) LP packing in C' to produce
</sup>

given by Example 6.6 (i), (ii) with the constructed $\binom{n/4}{4}$, 4) LP-packing in G' to produce the required $\binom{n}{4}$, 3) NLP-packing in $H \times G'$.

(*ii*) In the case that u = 1, and w and each of the s_i is zero, the result is given by Example 6.6 (*iii*). Otherwise, let $G' = \mathbb{Z}_3^{2u-2} \times \mathbb{Z}_9^w \times \prod_{i=3}^v \mathbb{Z}_{3^i}^{2s_i}$ and construct an $\left(\frac{n/3}{3}, 3\right)$ LP-packing in G' relative to a subgroup of order n/3. To do so when w is even, take p = 3 and $(s_1, s_2) = (u - 1, \frac{w}{2})$ and $j = m - 1 \ge 0$ in Corollary 5.4. To do so when w is odd, note that $\sigma(3, w) \ge 1$ and take p = 3 and $(s_1, s_2) = (u - 1, 0)$ and $j = m - 1 \ge 0$ in Corollary 5.7.

Now use Theorem 6.8 to combine the (1, 2) NLP-packing in \mathbb{Z}_3^2 given by Example 6.6 (*iii*) with the constructed $\left(\frac{n/3}{3}, 3\right)$ LP-packing in G' to produce the required $\left(\frac{n}{3}, 2\right)$ NLP-packing in G.

Remark 6.14. Using Remark 6.2 (*i*), we can rephrase Result 2.10 in terms of $(\frac{n}{t}, t - 1)$ NLP-packings in groups of order n^2 . As we now describe, we then see that Corollary 6.13 is significantly more general than Result 2.10.

Result 2.10 (i) and Corollary 6.13 (i) each construct NLP-packings with parameters $(\frac{n}{4}, 3)$. The former result is restricted to groups of the form $\mathbb{Z}_{2}^{4u} \times \mathbb{Z}_{4}^{2w} \times \prod_{i=2}^{v} \mathbb{Z}_{2^{2i}}^{4s_{2i}}$ for $u + w \ge 1$, whereas the latter result applies to the forms $\mathbb{Z}_{2}^{2u+4} \times \mathbb{Z}_{4}^{w} \times \prod_{i=3}^{v} \mathbb{Z}_{2^{i}}^{2s_{i}}$ and $\mathbb{Z}_{2}^{2u} \times \mathbb{Z}_{4}^{w+2} \times \prod_{i=3}^{v} \mathbb{Z}_{2^{i}}^{2s_{i}}$ for $u \neq 1$ and $w \notin \{1, 2, 3\}$ and each $s_{i} \neq 1$.

Similarly, Result 2.10 (*ii*) and Corollary 6.13 (*ii*) each construct NLP-packings with parameters $(\frac{n}{3}, 2)$. The former result is restricted to groups of the form $\mathbb{Z}_3^{2u+2} \times \prod_{i=1}^v \mathbb{Z}_{3^{2i}}^{2s_{2i}}$, whereas the latter result applies to the form $\mathbb{Z}_3^{2u} \times \mathbb{Z}_9^w \times \prod_{i=3}^v \mathbb{Z}_{3^{i}}^{2s_i}$ for $u \neq 0$ and $w \neq 1$.

7. Relationship with Strongly Regular Bent Functions

In this section, we examine how LP-packings and NLP-packings are related to the hyperbolic and elliptic strongly regular bent functions introduced by Chen and Polhill in [20].

Definition 7.1 ([20, Thm. 3.7]). Let G and H be abelian groups. Let f be a surjective function from G to H satisfying $f(1_G) = 1_H$, and let f^{-1} be the pre-image of f.

- (i) f is a hyperbolic strongly regular bent function if $f^{-1}(h)$ is a regular $\left(\sqrt{|G|}, \frac{\sqrt{|G|}}{|H|}\right)$ Latin square type PDS for each $h \neq 1_H$, and $f^{-1}(1_H) 1_G$ is a regular $\left(\sqrt{|G|}, \frac{\sqrt{|G|}}{|H|} + 1\right)$ Latin square type PDS.
- (*ii*) f is an *elliptic strongly regular bent function* if $f^{-1}(h)$ is a regular $\left(\sqrt{|G|}, \frac{\sqrt{|G|}}{|H|}\right)$ negative Latin square type PDS for each $h \neq 1_H$, and $f^{-1}(1_H) - 1_G$ is a regular $\left(\sqrt{|G|}, \frac{\sqrt{|G|}}{|H|} - 1\right)$ negative Latin square type PDS.

We can relate hyperbolic strongly regular bent functions to LP-packings, and elliptic strongly regular bent functions to NLP-packings.

Proposition 7.2. Let G be an abelian group of order t^2c^2 , let U be a subgroup of G of order tc, and let H be an abelian group of order t.

- (i) If there exists a (c,t) LP-packing in G relative to U, then there exists a hyperbolic strongly regular bent function from G to H.
- (*ii*) The existence of a (c, t 1) NLP-packing in G is equivalent to the existence of an elliptic strongly regular bent function from G to H.

Proof. Write $H = \{1_H, h_1, h_2, \dots, h_{t-1}\}$.

(i) Let $\{P_1, \ldots, P_t\}$ be a (c, t) LP-packing in G relative to U, and define the surjective function $f: G \to H$ by

$$f(g) = \begin{cases} h_i & \text{if } g \in P_i \text{ for } i \in \{1, \dots, t-1\}, \\ 1_H & \text{if } g \in P_t + U. \end{cases}$$

Then by Remark 3.10 (i), f is a hyperbolic strongly regular bent function.

(*ii*) Suppose firstly that $\{P_1, \ldots, P_{t-1}\}$ is a (c, t-1) NLP-packing in G, and define the surjective function $f: G \to H$ by

$$f(g) = \begin{cases} h_i & \text{if } g \in P_i, \\ 1_H & \text{otherwise.} \end{cases}$$

Then by Definition 6.1, f is an elliptic strongly regular bent function from G to H.

Combinatorial theory 27 (2020), #P00

Conversely, suppose that $f: G \to H$ is an elliptic strongly regular bent function, and define subsets P_1, \ldots, P_{t-1} of G by

$$P_i = f^{-1}(h_i)$$
 for $i = 1, \dots, t - 1$.

Then by Definition 6.1, $\{P_1, \ldots, P_{t-1}\}$ is a (c, t-1) NLP-packing in G from G to H.

Remark 7.3. Proposition 7.2 shows the equivalence of an NLP-packing and an elliptic strongly regular bent function, but not the equivalence of an LP-packing and a hyperbolic strongly regular bent function f. In order to establish the converse to Proposition 7.2 (i), we would need to identify an order $\sqrt{|G|}$ subgroup U of G contained in $f^{-1}(1_H)$ for which $f^{-1}(1_H)-U$ is a regular $(\sqrt{|G|}, \frac{\sqrt{|G|}}{|H|})$ Latin square type PDS in G. The additional conditions on f required to guarantee the existence of such a subgroup U could be substantial, as we discuss in Example 7.4 (i) below.

The following example uses strongly regular bent functions to construct collections of PDSs, and provides an alternative derivation of a special case of Corollary 6.13 (*ii*).

Example 7.4 ([86, Thm. 1, Cor. 3]). Let p be a prime, and let ζ_p be a primitive p^{th} root of unity. Let g be a function from \mathbb{F}_{p^n} to \mathbb{F}_p , and let Tr be the trace function from \mathbb{F}_{p^n} to \mathbb{F}_p . The Walsh transform $\mathcal{W}_g : \mathbb{F}_{p^n} \to \mathbb{C}$ of g is given by

$$\mathcal{W}_g(b) = \sum_{x \in \mathbb{F}_{p^n}} \zeta_p^{g(x) + \operatorname{Tr}(bx)}.$$

The function g is *bent* if $|\mathcal{W}_g(b)| = p^{\frac{n}{2}}$ for all $b \in \mathbb{F}_{p^n}$. A bent function g is *weakly regular* if there exists a function $g^* : \mathbb{F}_{p^n} \to \mathbb{F}_p$ satisfying

 $\mathcal{W}_g(b) = \mu p^{\frac{n}{2}} \zeta_p^{g^*(b)}$ for some $\mu \in \mathbb{C}$ with $|\mu| = 1$ and for all $b \in \mathbb{F}_{p^n}$,

and is *regular* (a much stronger condition) if there exists a function $g^* : \mathbb{F}_{p^n} \to \mathbb{F}_p$ such that $\mathcal{W}_g(b) = p^{\frac{n}{2}} \zeta_p^{g^*(b)}$ for all $b \in \mathbb{F}_{p^n}$.

Suppose $f : \mathbb{F}_{3^{2s}} \to \mathbb{F}_3$ is a weakly regular bent function satisfying f(-x) = f(x) and f(0) = 0. Then exactly one of the following holds:

- (i) $|f^{-1}(0)| = 3^{2s-1} + 2 \cdot 3^{s-1}$, and f is a hyperbolic strongly regular bent function from $\mathbb{F}_{3^{2s}}$ to \mathbb{F}_3 : each of $f^{-1}(1)$ and $f^{-1}(2)$ is a regular $(3^s, 3^{s-1})$ Latin square type PDS in $\mathbb{F}_{3^{2s}}$, and $f^{-1}(0) \setminus \{0\}$ is a regular $(3^s, 3^{s-1} + 1)$ Latin square type PDS in $\mathbb{F}_{3^{2s}}$.
- (*ii*) $|f^{-1}(0)| = 3^{2s-1} 2 \cdot 3^{s-1}$, and f is an elliptic strongly regular bent function from $\mathbb{F}_{3^{2s}}$ to \mathbb{F}_3 : each of $f^{-1}(1)$ and $f^{-1}(2)$ is a regular $(3^s, 3^{s-1})$ negative Latin square type PDS in $\mathbb{F}_{3^{2s}}$, and $f^{-1}(0) \setminus \{0\}$ is a regular $(3^s, 3^{s-1} 1)$ negative Latin square type PDS in $\mathbb{F}_{3^{2s}}$; equivalently, $\{f^{-1}(1), f^{-1}(2)\}$ is a $(3^{s-1}, 2)$ NLP-packing in $(\mathbb{F}_{3^{2s}}, +) = \mathbb{Z}_3^{2s}$.

Following Remark 7.3, we mention that in order to obtain a $(3^{s-1}, 3)$ LP-packing in the group $(\mathbb{F}_3^{2s}, +) = \mathbb{Z}_3^{2s}$ relative to \mathbb{Z}_3^s from the hyperbolic strongly regular bent function f of (i), we would firstly need to identify an order 3^s subgroup U of \mathbb{Z}_3^{2s} contained in $f^{-1}(0)$, and secondly require $f^{-1}(0) - U$ to be a regular $(3^s, 3^{s-1})$ Latin square type PDS in \mathbb{Z}_3^{2s} . The first condition alone requires f to be normal [16, p. 106], which in combination with the property that f is weakly regular requires in turn that f should be regular [16, Thm. 6]. This demonstrates that, in general, an LP-packing is a significantly more constrained structure than a hyperbolic strongly regular bent function.

Remark 7.5.

- (i) In [20, Lemma 3.15], an incomplete attempt was made to lift a hyperbolic strongly regular bent function on \mathbb{Z}_p^{2a} to a hyperbolic strongly regular bent function on $\mathbb{Z}_{p^2}^{2a}$. We were able to accomplish this lifting in Theorem 5.3, using the more constrained structure of an LPpacking. This underscores the importance of identifying the role of the subgroup U in the definition of an LP-packing.
- (*ii*) Corollaries 5.4, 5.7 and 6.13 recover the hyperbolic and elliptic strongly regular bent functions of [20, Prop. 3.13] and strengthen those of [20, Prop. 3.14]. The hyperbolic and elliptic strongly regular bent functions constructed in [20, Prop. 3.12] have codomain of size 2, and are equivalent to reversible Hadamard difference sets. We examine this connection in more detail in Section 8 below.

8. Relationship with reversible Hadamard difference sets

In this section, we examine how LP-packings and NLP-packings are closely related to reversible Hadamard difference sets.

Definition 8.1. Let G be a group G of order v. A (v, k, λ) -difference set in G is a (v, k, λ, λ) partial difference set in G. A Hadamard difference set has parameters $(v, k, \lambda) = (4c^2, c(2c - 1), c(c - 1))$ for some integer c. A difference set D is reversible if $D = D^{(-1)}$.

By Definition 2.1, a reversible (v, k, λ) -difference set that does not contain 1_G is the same as a regular (v, k, λ, λ) partial difference set. The following result is a direct consequence of Definition 6.1 and Lemmas 2.3 and 2.5.

Lemma 8.2. Let c be a positive integer, and let P be a c(2c - 1)-subset of an abelian group G of order $4c^2$. Then the following statements are equivalent.

- (i) P is a $(4c^2, c(2c-1), c(c-1))$ reversible Hadamard difference set in G, and $1_G \notin P$
- (*ii*) $\chi(P) \in \{-c, c\}$ for all nonprincipal characters χ of G
- (iii) $\{G P\}$ is a (c, 1) NLP-packing in G.

The next result follows from Definition 3.1 and Lemmas 2.5 and 8.2.

Combinatorial theory 27 (2020), #P00

Lemma 8.3. Let c be a positive integer, let G be an abelian group of order $4c^2$, and let U be a subgroup of order 2c. Then P_1 , P_2 are disjoint $(4c^2, c(2c-1), c(c-1))$ reversible Hadamard difference sets in G whose union is G - U if and only if $\{P_1, P_2\}$ is a (c, 2) LP-packing in G relative to U.

We now use LP-packings in 2-groups to produce infinite families of reversible Hadamard difference sets.

Corollary 8.4. Let u, w and s_3, \ldots, s_v be nonnegative integers (not all zero). For each $i = 3, \ldots, v$, let $G_i = \mathbb{Z}_{2^i}^{2s_i}$ and let U_i be a subgroup of G_i of order 2^{is_i} . Let $G = \mathbb{Z}_2^{2u} \times \mathbb{Z}_4^w \times \prod_{i=3}^v G_i$, let $U = \mathbb{Z}_2^u \times (2\mathbb{Z}_4)^w \times \prod_{i=3}^v U_i$, and let $|G| = 4c^2$. Then

- (i) for $w \neq 1$, there exist two disjoint $(4c^2, c(2c-1), c(c-1))$ reversible Hadamard difference sets in G whose union is G U.
- (*ii*) there exists a $(4c^2, c(2c-1), c(c-1))$ reversible Hadamard difference set in G.

Proof.

- (i) By Lemma 8.3, an equivalent statement is that for $w \neq 1$ there exists a (c, 2) LP-packing in G relative to U. To show this when w is even, take p = 2 and $(s_1, s_2) = (u, \frac{w}{2})$ and $j = m - 1 \ge 0$ in Corollary 5.4. To show this when w is odd, note that $\sigma(2, w) = \frac{w - 1}{2} \ge 1$ and take p = 2 and $(s_1, s_2) = (u, 0)$ and $j = m - 1 \ge 0$ in Corollary 5.7.
- (*ii*) In view of part (*i*), we may assume w = 1. In the case that u and each of the s_i is zero, there is a trivial reversible Hadamard difference set D in \mathbb{Z}_4 comprising just the identity element. Otherwise, take p = 2 and $(s_1, s_2) = (u, 0)$ and $j = m 1 \ge 0$ in Corollary 5.4 to produce a $(\frac{c}{2}, 2)$ LP-packing in $\mathbb{Z}_2^{2u} \times \prod_{i=3}^{v} G_i$ relative to some subgroup of order c. Use Theorem 6.8 to combine this with the (1, 1) NLP-packing $\{\mathbb{Z}_4 D\}$ in \mathbb{Z}_4 (as given by Lemma 8.2) to produce a (c, 1) NLP-packing in G, then apply Lemma 8.2.

Remark 8.5.

- (i) Corollary 8.4 (i) significantly extends a previous construction for $G = \mathbb{Z}_{2^a}^2$ and noncyclic U of order 2^a [36, Thm. 3.1].
- (ii) All abelian 2-groups in which a reversible Hadamard difference set is known to exist are recovered by Corollary 8.4 (ii); several other families of abelian groups are also known to contain reversible Hadamard difference sets (see [7, Chap. VI, Sect. 14]), and all these groups are recovered by [20, Prop. 3.12]) using elliptic strongly regular bent functions.
- (*iii*) The construction of linking systems of reversible Hadamard difference sets in [35, Thm. 5.3] depends on the existence of collections of disjoint partial difference sets in abelian 2-groups satisfying mutual structural properties. After some modification, we can reinterpret this construction as combining a $(2^{(a-1)s}, 2^s)$ LP-packing in an abelian group of order 2^{2as} , relative to some subgroup of order 2^{as} , with the complement of a $(2^s 1, 2^{s-1} 1, 2^{s-2} 1)$ -difference set in $\mathbb{Z}_{2^{s-1}}$ [7, Chap. VI, Thm. 1.10] for s > 1 and a > 0.

9. Open Problems

In this section, we propose some open problems for future research.

- P1. The recursive constructions of LP-packings described in Section 5 rest on two base cases: a $(1, p^s)$ LP-packing in \mathbb{Z}_p^{2s} relative to \mathbb{Z}_p^s derived from a spread (Example 3.7), and the LPpackings in *p*-groups of odd rank given by Result 5.6. Find more base cases, especially in groups of odd rank (when a spread does not exist).
- P2. Section 6 combines small examples of NLP-packings with the constructed families of LPpackings to produce families of NLP-packings. Find new small examples of NLP-packings.
- P3. All known (c, t) LP-packings with t > 2, and all known (c, t) NLP-packings with t > 1, occur in *p*-groups (see Section 8 for discussion of (c, 2) LP-packings and (c, 1) NLP-packings.) Find examples in groups other than *p*-groups.
- P4. Nonexistence results for LP-packings and NLP-packings can be obtained from known nonexistence results for single PDSs (see [88], for example). We can also obtain nonexistence results for these packings from known nonexistence results for reversible Hadamard difference sets [7, Chap. VI, Sect. 14], by applying Lemmas 3.11 and 8.3 to (c, 2t) LP-packings, and Lemmas 6.7 and 8.2 to (c, 2t-1) NLP-packings. Establish further nonexistence results for LP-packings and NLP-packings.
- P5. Construct hyperbolic strongly regular bent functions satisfying additional conditions in order to produce LP-packings in new groups (see Remark 7.3).
- P6. Are there collections of partial difference sets with advantageous structure, other than LPpackings and NLP-packings, that can be analyzed using methods similar to those of this paper?

Acknowledgements

The authors gratefully acknowledge helpful discussions with Samuel Simon at the outset of this research. We thank Alexander Pott for his hospitality when the first author visited the second at Otto von Guericke University, Magdeburg in Summer 2018 to pursue this research. We thank the anonymous reviewers for their detailed and helpful comments.

References

- [1] A. Aguglia. Quasi-Hermitian varieties in $PG(r, q^2)$, q even. Contrib. Discrete Math., 8(1):31–37, 2013. MR 3118898
- [2] A. Aguglia, A. Cossidente, and G. Korchmáros. On quasi-Hermitian varieties. J. Combin. Des., 20(10):433–447, 2012. MR 2959282

- [3] R.A. Bailey and D. Jungnickel. Translation nets and fixed-point-free group automorphisms. *J. Combin. Theory Ser. A*, 55(1):1–13, 1990. MR 1070011
- [4] J. Bamberg, S. Kelly, M. Law, and T. Penttila. Tight sets and *m*-ovoids of finite polar spaces. *J. Combin. Theory Ser. A*, 114(7):1293–1314, 2007. MR 2353124
- [5] J. Bamberg, M. Lee, K. Momihara, and Q. Xiang. A new infinite family of hemisystems of the Hermitian surface. *Combinatorica*, 38(1):43–66, 2018. MR 3776347
- [6] L.D. Baumert, W.H. Mills, and R.L. Ward. Uniform cyclotomy. *J. Number Theory*, 14(1):67–82, 1982. MR 0644901
- [7] T. Beth, D. Jungnickel, and H. Lenz. Design Theory. Vol. I, volume 69 of Encyclopedia of Mathematics and its Applications. Cambridge University Press, Cambridge, second edition, 1999. MR 1729456
- [8] J. De Beule, J. Demeyer, K. Metsch, and M. Rodgers. A new family of tight sets in $Q^+(5, q)$. Des. Codes Cryptogr., 78(3):655–678, 2016. MR 3459217
- [9] A. Blokhuis and M. Lavrauw. On two-intersection sets with respect to hyperplanes in projective spaces. J. Combin. Theory Ser. A, 99(2):377–382, 2002. MR 1917029
- [10] A.E. Brouwer and H. van Maldeghem. Strongly regular graphs. https://homepages. cwi.nl/~aeb/math/srg/rk3/srgw.pdf
- [11] A.E. Brouwer, R.M. Wilson, and Q. Xiang. Cyclotomy and strongly regular graphs. J. Algebr. Comb., 10(1):25–28, 1999. MR 1701281
- [12] B. De Bruyn. On some 2-tight sets of polar spaces. Ars Combin., 133:115–131, 2017. MR 3699834
- [13] R. Calderbank and W.M. Kantor. The geometry of two-weight codes. *Bull. London Math.* Soc., 18(2):97–122, 1986. MR 0818812
- [14] C. Carlet, G. Gong, and Y. Tan. Quadratic zero-difference balanced functions, APN functions and strongly regular graphs. *Des. Codes Cryptogr.*, 78(3):629–654, 2016. MR 3459216
- [15] A. Çeşmelioğlu and W. Meidl. Bent and vectorial bent functions, partial difference sets, and strongly regular graphs. *Adv. Math. Commun.*, 12(4):691–705, 2018. MR 3917637
- [16] A. Çeşmelioğlu, W. Meidl, and A. Pott. Generalized Maiorana-McFarland class and normality of *p*-ary bent functions. *Finite Fields Appl.*, 24:105–117, 2013. MR 3093860
- [17] Y.M. Chee, Y. Tan, and X.D. Zhang. Strongly regular graphs constructed from *p*-ary bent functions. *J. Algebr. Comb.*, 34(2):251–266, 2011. MR 2811149
- [18] Y.Q. Chen and T. Feng. Abelian and non-abelian Paley type group schemes. Des. Codes Cryptogr., 68(1-3):141–154, 2013. MR 3046342
- [19] Y.Q. Chen and J. Polhill. Paley type group schemes and planar Dembowski-Ostrom polynomials. *Discrete Math.*, 311(14):1349–1364, 2011. MR 2795545
- [20] Y.Q. Chen and J. Polhill. Partial difference sets and amorphic group schemes from pseudoquadratic bent functions. *J. Algebr. Comb.*, 37(1):11–26, 2013. MR 3016299

- [21] Y.Q. Chen, D.K. Ray-Chaudhuri, and Q. Xiang. Constructions of partial difference sets and relative difference sets using Galois rings. II. J. Combin. Theory Ser. A, 76(2):179–196, 1996. MR 1416013
- [22] F. De Clerck, N. Hamilton, C.M. O'Keefe, and T. Penttila. Quasi-quadrics and related structures. Australas. J. Combin., 22:151–166, 2000. MR 1795332
- [23] A. Cossidente. Combinatorial structures in finite classical polar spaces. In Surveys in Combinatorics 2017, volume 440 of London Math. Soc. Lecture Note Ser., pages 204–237. Cambridge Univ. Press, Cambridge, 2017. MR 3728108
- [24] A. Cossidente, N. Durante, G. Marino, T. Penttila, and A. Siciliano. The geometry of some two-character sets. *Des. Codes Cryptogr.*, 46(2):231–241, 2008. MR 2368996
- [25] A. Cossidente and H. Van Maldeghem. The simple exceptional group $G_2(q)$, q even, and two-character sets. J. Combin. Theory Ser. A, 114(5):964–969, 2007. MR 2333144
- [26] A. Cossidente and G. Marino. Veronese embedding and two-character sets. Des. Codes Cryptogr., 42(1):103–107, 2007. MR 2277061
- [27] A. Cossidente, G. Marino, and F. Pavese. Projective Paley sets. J. Combin. Des., 27(10):598–613, 2019. MR 3998644
- [28] A. Cossidente and F. Pavese. Intriguing sets of $\mathcal{W}(5,q)$, q even. J. Combin. Theory Ser. A, 127:303–313, 2014. MR 3252665
- [29] A. Cossidente and F. Pavese. Intriguing sets of quadrics in PG(5,q). Adv. Geom., 17(3):339–345, 2017. MR 3680356
- [30] A. Cossidente and F. Pavese. On intriguing sets of finite symplectic spaces. *Des. Codes Cryptogr.*, 86(5):1161–1174, 2018. MR 3779049
- [31] A. Cossidente and T. Penttila. Two-character sets arising from gluings of orbits. *Graphs Combin.*, 29(3):399–406, 2013. MR 3053589
- [32] J.A. Davis. Partial difference sets in *p*-groups. *Arch. Math.*, 63(2):103–110, 1994. MR 1289290
- [33] J.A. Davis and J. Jedwab. A unifying construction for difference sets. J. Combin. Theory Ser. A, 80(1):13–78, 1997. MR 1472105
- [34] J.A. Davis and J. Jedwab. A unified approach to difference sets with gcd(v, n) > 1. In A. Pott et al., editors, *Difference Sets, Sequences and Their Correlation Properties*, volume 542 of *NATO Science Series C*, pages 85–112. Kluwer Academic Publishers, Dordrecht, 1999. MR 1735394
- [35] J.A. Davis, W.J. Martin, and J.B. Polhill. Linking systems in nonelementary abelian groups. J. Combin. Theory Ser. A, 123(1):92–103, 2014. MR 3157802
- [36] J.A. Davis and J. Polhill. Difference set constructions of DRADs and association schemes. J. Combin. Theory Ser. A, 117(5):598–605, 2010. MR 2600978
- [37] J.A. Davis and Q. Xiang. A family of partial difference sets with Denniston parameters in nonelementary abelian 2-groups. *European J. Combin.*, 21(8):981–988, 2000. MR 1797679

- [38] J.A. Davis and Q. Xiang. Negative Latin square type partial difference sets in nonelementary abelian 2-groups. J. London Math. Soc. (2), 70(1):125–141, 2004. MR 2064754
- [39] J.A. Davis and Q. Xiang. Amorphic association schemes with negative Latin square-type graphs. *Finite Fields Appl.*, 12(4):595–612, 2006. MR 2257085
- [40] J.F. Dillon. *Elementary Hadamard difference sets*. Ph. D. Thesis, University of Maryland, 1974. MR 2624542
- [41] T. Etzion and L. Storme. Galois geometries and coding theory. *Des. Codes Cryptogr.*, 78(1):311–350, 2016. MR 3440233
- [42] T. Feng, K. Momihara, and Q. Xiang. Cameron-Liebler line classes with parameter $x = \frac{q^2-1}{2}$. J. Combin. Theory Ser. A, 133:307–338, 2015. MR 3325637
- [43] T. Feng, K. Momihara, and Q. Xiang. Constructions of strongly regular Cayley graphs and skew Hadamard difference sets from cyclotomic classes. *Combinatorica*, 35(4):413–434, 2015. MR 3386052
- [44] T. Feng, B. Wen, Q. Xiang, and J. Yin. Partial difference sets from quadratic forms and p-ary weakly regular bent functions. In *Number Theory and Related Areas*, volume 27 of *Adv. Lect. Math. (ALM)*, pages 25–40. Int. Press, Somerville, MA, 2013. MR 3185869
- [45] T. Feng and Q. Xiang. Strongly regular graphs from unions of cyclotomic classes. J. Combin. Theory Ser. B, 102(4):982–995, 2012. MR 2927417
- [46] G.A. Fernández-Alcober, R. Kwashira, and L. Martínez. Cyclotomy over products of finite fields and combinatorial applications. *European J. Combin.*, 31(6):1520–1538, 2010. MR 2660401
- [47] N. Hamilton and T. Penttila. Sets of type (a, b) from subgroups of $\Gamma L(1, p^R)$. J. Algebr. Combin., 13(1):67–76, 2001. MR 1817704
- [48] X.-D. Hou. Bent functions, partial difference sets, and quasi-Frobenius local rings. Des. Codes Cryptogr., 20(3):251–268, 2000. MR 1779309
- [49] X.-D. Hou. New partial difference sets in *p*-groups. J. Combin. Des., 10(6):394–402, 2002.
 MR 1931798
- [50] X.-D. Hou. Rings and constructions of partial difference sets. *Discrete Math.*, 270(1-3):149–176, 2003. MR 1997895
- [51] X.-D. Hou, K.H. Leung, and Q. Xiang. New partial difference sets in $\mathbb{Z}_{p^2}^t$ and a related problem about Galois rings. *Finite Fields Appl.*, 7(1):165–188, 2001. MR 1803942
- [52] X.-D. Hou and A.A. Nechaev. A construction of finite Frobenius rings and its application to partial difference sets. J. Algebra, 309(1):1–9, 2007. MR 2301229
- [53] D.R. Hughes and F.C. Piper. Projective Planes, volume 6 of Graduate Texts in Mathematics. Springer-Verlag, New York-Berlin, 1973. MR 0333959
- [54] S. Innamorati, M. Zannetti, and F. Zuanni. On two character $(q^7 + q^5 + q^2 + 1)$ -sets in PG(4, q^2). J. Geom., 106(2):287–296, 2015. MR 3353837
- [55] D. Jungnickel. Existence results for translation nets, II. J. Algebra, 122(2):288–298, 1989. MR 0999074

- [56] D. Jungnickel. Partial spreads over \mathbb{Z}_q . Linear Algebra Appl., 114:95–102, 1989. MR 0986867
- [57] K.H. Leung and S.L. Ma. Constructions of partial difference sets and relative difference sets on *p*-groups. *Bull. Lond. Math. Soc.*, 22(6):533–539, 1990. MR 1099002
- [58] K.H. Leung and S.L. Ma. Partial difference sets with Paley parameters. Bull. London Math. Soc., 27(6):553–564, 1995. MR 1348709
- [59] K.H. Leung and S.L. Ma. A construction of partial difference sets in $\mathbb{Z}_{p^2} \times \mathbb{Z}_{p^2} \times \cdots \times \mathbb{Z}_{p^2}$. Des. Codes Cryptogr., 8(1-2):167–172, 1996. MR 1393982
- [60] S.L. Ma. A survey of partial difference sets. Des. Codes Cryptogr., 4(3):221–261, 1994. MR 1277942
- [61] M.E. Malandro and K.W. Smith. Partial difference sets in $C_{2^n} \times C_{2^n}$. Discrete Math., 343(4):111744, 2020. MR 4045555
- [62] K. Momihara. Certain strongly regular Cayley graphs on $\mathbb{F}_{2^{2(2s+1)}}$ from cyclotomy. *Finite Fields Appl.*, 25:280–292, 2014. MR 3130604
- [63] K. Momihara. Construction of strongly regular Cayley graphs based on three-valued Gauss periods. *European J. Combin.*, 70:232–250, 2018. MR 3779616
- [64] K. Momihara. New constructions of strongly regular Cayley graphs on abelian non pgroups. J. Combin. Theory Ser. A, 184:Paper No. 105514, 20, 2021. MR 4293037
- [65] K. Momihara and Y. Okumura. New strongly regular decompositions of the complete graphs with prime power vertices. *Finite Fields Appl.*, 36:63–80, 2015. MR 3396376
- [66] K. Momihara, Q. Wang, and Q. Xiang. Cyclotomy, difference sets, sequences with low correlation, strongly regular graphs, and related geometric substructures. In K.-U. Schmidt and A. Winterhof, editors, *Proceedings of the workshop "Pseudo-Randomness and Finite Fields"*, RICAM, Linz, Austria, 2018.
- [67] K. Momihara and Q. Xiang. Lifting constructions of strongly regular Cayley graphs. *Finite Fields Appl.*, 26:86–99, 2014. MR 3151359
- [68] K. Momihara and Q. Xiang. Strongly regular Cayley graphs from partitions of subdifference sets of the Singer difference sets. *Finite Fields Appl.*, 50:222–250, 2018. MR 3759792
- [69] U. Ott. Families of cyclotomic partial difference sets from a given partial difference set. researchgate.net/publication/317663576, 2016.
- [70] U. Ott. On Jacobi sums, difference sets and partial difference sets in Galois domains. Des. Codes Cryptogr., 80(2):241–281, 2016. MR 3513887
- [71] U. Ott. Some new families of partial difference sets in finite fields. J. Geom., 107(2):267–278, 2016. MR 3519948
- [72] F. Ozbudak and R.M. Pelen. Strongly regular graphs arising from non-weakly regular bent functions. *Cryptogr. Commun.*, 11(6):1297–1306, 2019. MR 4033176
- [73] F. Pavese. Geometric constructions of two-character sets. *Discrete Math.*, 338(3):202–208, 2015. MR 3291885

- [74] J. Polhill. Generalizations of partial difference sets from cyclotomy to nonelementary abelian *p*-groups. *Electron. J. Combin.*, 15:#R125, 13 pages, 2008. MR 2443140
- [75] J. Polhill. Negative Latin square type partial difference sets and amorphic association schemes with Galois rings. *J. Combin. Des.*, 17(3):266–282, 2009. MR 2512868
- [76] J. Polhill. Paley type partial difference sets in non *p*-groups. *Des. Codes Cryptogr.*, 52(2):163–169, 2009. MR 2496253
- [77] J. Polhill. Paley partial difference sets in groups of order n^4 and $9n^4$ for any odd n > 1. J. *Combin. Theory Ser. A*, 117(8):1027–1036, 2010. MR 2677671
- [78] J. Polhill. A new family of partial difference sets in 3-groups. *Des. Codes Cryptogr.*, 87(7):1639–1646, 2019. MR 3954423
- [79] J. Polhill, J.A. Davis, and K. Smith. A new product construction for partial difference sets. *Des. Codes Cryptogr.*, 68(1-3):155–161, 2013. MR 3046343
- [80] J.B. Polhill. Constructions of nested partial difference sets with Galois rings. Des. Codes Cryptogr., 25(3):299–309, 2002. MR 1900775
- [81] J.B. Polhill. A construction of layered relative difference sets using Galois rings. Ars Combin., 78:83–94, 2006. MR 2194752
- [82] J.B. Polhill. New negative Latin square type partial difference sets in nonelementary abelian 2-groups and 3-groups. *Des. Codes Cryptogr.*, 46(3):365–377, 2008. MR 2372845
- [83] A. Pott. Finite Geometry and Character Theory, volume 1601 of Lecture Notes in Mathematics. Springer-Verlag, Berlin, 1995. MR 1440858
- [84] D.K. Ray-Chaudhuri and Q. Xiang. Constructions of partial difference sets and relative difference sets using Galois rings. *Des. Codes Cryptogr.*, 8(1-2):215–227, 1996. MR 1393986
- [85] B. Schmidt and C. White. All two-weight irreducible cyclic codes? *Finite Fields Appl.*, 8(1):1–17, 2002. MR 1872788
- [86] Y. Tan, A. Pott, and T. Feng. Strongly regular graphs associated with ternary bent functions. J. Combin. Theory Ser. A, 117(6):668–682, 2010. MR 2645184
- [87] E. R. van Dam and M. Muzychuk. Some implications on amorphic association schemes. *J. Combin. Theory Ser. A*, 117(2):111–127, 2010. MR 2557883
- [88] Z. Wang. New necessary conditions on (negative) Latin square type partial difference sets in abelian groups. J. Combin. Theory Ser. A, 172:105208, 10, 2020. MR 4050735
- [89] G. Weng, W. Qiu, Z. Wang, and Q. Xiang. Pseudo-Paley graphs and skew Hadamard difference sets from presemifields. *Des. Codes Cryptogr.*, 44(1-3):49–62, 2007. MR 2336393
- [90] H. Wielandt. Zur theorie der einfach transitiven permutationsgruppen. *Math. Z.*, 40(1):582–587, 1936. MR 1545582
- [91] Q. Xiang and J.A. Davis. Constructions of low rank relative difference sets in 2-groups using Galois rings. *Finite Fields Appl.*, 6(2):130–145, 2000. MR 1755762